HOMOLOGICAL VECTOR FIELDS OVER DIFFERENTIABLE STACKS

DANIEL ÁLVAREZ AND MIQUEL CUECA

ABSTRACT. In this work we solve the problem of providing a Morita invariant definition of Lie and Courant algebroids over Lie groupoids. By relying on supergeometry, we view these structures as instances of vector fields on graded groupoids which are homological up to homotopy. We describe such vector fields in general from two complementary viewpoints: firstly, as Maurer-Cartan elements in a differential graded Lie algebra of multivector fields and, secondly, we also view them from a categorical approach, in terms of functors and natural transformations. Thereby, we obtain a unifying conceptual framework for studying LA-groupoids, L_2 -algebroids (including semistrict Lie 2-algebras and 2-term representations up to homotopy), infinitesimal gerbe prequantizations, higher gauge theory (specifically, 2-connections on 2-bundles), quasi-Poisson groupoids and (twisted) multiplicative Courant algebroids.

Contents

1. Introduction	2
1.1. Summary	2 2
1.2. Future outlook	3
2. Quasi Q-groupoids	д С
2.1. Graded manifolds	6
2.2. Lie groupoids and algebroids in the category of graded ma	anifolds 7
2.3. Homological vector fields on graded Lie groupoids	8
2.3.1. Interpretation in terms of natural transformations	ç
2.3.2. Interpretation in terms of the Maurer-Cartan equation	n 10
2.4. Quasi QP-groupoids	12
2.5. Morphisms of quasi Q-groupoids	13
2.5.1. 1-morphisms	13
2.5.2. Composition of 1-morphisms	15
2.5.3. 2-morphisms	15
3. Degree one quasi Q-groupoids: quasi LA-groupoids	15
3.1. The functor $[1]$	16
3.1.1. Vector fields on 1-manifolds	16
3.1.2. LA-groupoids as Q-groupoids	17
3.2. Quasi LA-groupoids	18
3.2.1. Quasi PVB-groupoids and duality	19
3.2.2. Lie theory of quasi LA-groupoids	19
3.3. L_2 -algebroids	21
3.4. The category of multiplicative sections	24
3.5. Lie bialgebroids over Lie groupoids	26

²⁰²⁰ Mathematics Subject Classification. 18F20, 58A50, 53D17, 58H05.

Key words and phrases. Differentiable stacks, Homological vector fields, Lie algebroids.

DANIEL ÁLVAREZ AND MIQUEL CUECA

3.6. Examples	27
3.6.1. Cotangent bundles of quasi Poisson groupoids	27
3.6.2. Central extensions and prequantum algebroids	28
3.6.3. L_2 -algebra actions on Lie groupoids	30
3.6.4. 2-term representations up to homotopy	31
3.6.5. Atiyah algebroids of 2-bundles and 2-connections	32
4. Degree two quasi Q-groupoids: quasi CA-groupoids	34
4.1. Roytenberg-Severa correspondence	34
4.2. Degree 2 symplectic quasi Q-groupoids	35
5. Epilogue: Lie and Courant algebroids through the stacky glasses	37
Appendix A. Double quasi Poisson groupoids	38
A.1. Quasi Poisson LA-groupoids and multiplicative Lie quasi-bialgebroids	40
A.2. Integration of quasi Poisson LA-groupoids over a unit groupoid	40
Appendix B. Infinitesimal derivations and linear sections	41
References	43

1. INTRODUCTION

It is usually desirable in both mathematics and physics to describe geometric structures in a flexible enough manner that they can be transported under certain change of frame of reference operations. More specifically, whenever we want to define a geometric structure over a differentiable stack, understood as a class of *weakly equivalent* Lie groupoids [13], we have to make sure it is such that it can be transported under Morita equivalences. It has been realized that the appropriate way to do so is by providing definitions that rely on homotopies or natural isomorphisms instead of equalities. In this context, the program of *categorifying* several fundamental concepts is particularly relevant: see [10, 9, 8, 11, 16, 96] for a discussion of the ideas of Lie group, Lie algebra, principal bundle, parallel transport and Poisson and symplectic manifolds in this spirit.

For Lie algebras, in particular, two generalizations naturally appeared: (1) one can define skew-symmetric brackets that satisfy the Jacobi identity up to chain homotopy in a complex of vector spaces, this is the idea behind L_{∞} -algebras; (2) on the other hand, we can think of a categorified Lie algebra as a category itself which is endowed with a Lie bracket that is a functor and that satisfies the Jacobi identity up to a natural isomorphism. Both approaches were illustrated and contrasted in [9].

In this work we further develop this program by examining the idea of *Lie algebroid*. Defining Lie algebroids up to homotopy as L_2 -algebroids (or split degree 2 *Q*-manifolds) is straightforward, see for instance [15, 88]. On the other hand, a partial categorification of Lie algebroids was achieved by the introduction of LA-groupoids [63]; however, the properties of such objects are still characterized by equalities instead of natural isomorphisms and so they are not Morita invariant. Notwithstanding this fact, there have been some attempts to use LA-groupoids to define Lie algebroids over differentiable stacks [83, 94] (see also [14] for a study of Lie algebroids and stacks in the algebraic category).

The goal of this work is to unify the study of those two classes of objects in a way that sheds light on many concrete examples that have appeared in the literature. In doing so we aim to give a framework which is concrete enough to perform explicit computations and yet is sufficiently general to explain and put into a suitable context a number of results and constructions about

 $\mathbf{2}$

homotopic or categorified notions in classical differential geometry. Our perspective encompasses and clarifies diverse notions of Lie bialgebroids [12, 24, 36, 55], the emergence of Lie 2-algebra structures on the spaces of multiplicative sections of LA-groupoids and quasi Poisson groupoids [34, 57, 80], the definition of a flat 2-connection on a 2-bundle in higher gauge theory [8, 11, 18], and Courant algebroids over groupoids [2, 72, 79] (possibly twisted as in [45]).

Our interpretation of L_2 -algebroids in categorical terms sheds light on the integration of 2term representations up to homotopy via 2-functors [17] (the version of Lie's second theorem relevant for this interpretation is discussed in [97]). We plan to examine these global aspects of the theory further in future work.

1.1. Summary. Due to the length of this work, we include here a brief description of our contributions. Over the last few decades the language of supergeometry has led to the unified study of many different geometric structures such as Poisson manifolds [86], Lie algebroids [90], Lie bialgebroids and Courant algebroids [85]; all these objects can be viewed as Q-manifolds, i.e. N-graded manifolds endowed with a vector field Q that squares to zero; in this situation, Q is called a Q-structure or a homological vector field.

It has long been realized that Lie groupoids are very convenient models for differentiable stacks, see [13] and compare it with the original algebraic approach of [6]. In the light of this, we can define a graded differentiable stack, i.e. a smooth stack in the category of \mathbb{N} -graded manifolds, as the Morita equivalence class of a graded Lie groupoid as in [70, 72]. So the first question that we address is the following

(Q1) What is a homological vector field on a graded differentiable stack?

We give an answer to this question in §2 by recalling the general definition of graded groupoids as in [70, 72] and introducing degree 1 multiplicative vector fields Q over them that are homological *up to homotopy* in the sense that Q^2 is equal to a simplicially exact term. This is in contrast with the Q-groupoids introduced in [70, 72] where Q^2 is required to be equal to 0. The resulting condition immediately characterizes Q as part of a Maurer-Cartan element in the dgla of multiplicative multivector fields on a groupoid (see §2.3.2) and hence we get that it is a structure that can be transported along Morita equivalences, providing thus the appropriate concept of Q-structure over a differentiable stack. We call these objects *quasi Q-groupoids* (Definition 2.8).

We also characterize the above Maurer-Cartan elements as natural transformations in §2.3.1. This equivalent viewpoint allows us to view these objects as genuine categorifications of Q-manifolds, in the same way that Lie 2-algebras are a categorification of Lie algebras [9]. This categorical viewpoint naturally leads to our definition of morphism between quasi Q-groupoids in §2.5. Let us just point out that this concept extends the idea of gauge transformation between Maurer-Cartan elements and includes morphisms of semistrict Lie 2-algebras [9] as particular examples.

Poisson structures have been frequently studied in the presence of a homological vector field, most famously under the guise of QP-manifolds [1]. So another closely related question to (Q1) is the following:

(Q2) What is a *QP* manifold structure on a graded differentiable stack?

We give an answer to this question in Definition 2.19, drawing inspiration from homotopy and shifted Poisson structures [16, 73, 81]. We pay particular emphasis to non degenerate Poisson structures as that situation is what allows us to consider some of the most interesting examples as we shall see below.

In the rest of the paper we specialize our answers to (Q1) and (Q2) in many situations. Since Lie algebroid structures are equivalent to degree 1 *Q*-manifolds [90], for graded groupoids of degree 1, (Q1) becomes the following question:

(Q3) What is a Lie algebroid over a differentiable stack?

Our answer to this question in §3 specializes, on one hand, to LA-groupoids [63] viewed as Q-groupoids [70] and, on the other, to L_2 -algebroids, see Theorem 3.15. We refer to the categorified Lie algebroids thus obtained as *quasi LA-groupoids*. Our viewpoint allows us to consider many examples: in fact, we show that quasi LA-groupoids appear as central extensions determined by closed 1-shifted 2-forms and quasi Poisson groupoids structures, see §3.6.2. Furthermore, we explain how 2-term representations up to homotopy §3.6.4 and flat 2-connections are examples of quasi LA-groupoid morphisms §3.6.5.

Lie algebroid structures are in duality with respect to linear Poisson structures. In §3.2.1 we promote that relationship to a duality between our quasi LA-groupoids and linear quasi Poisson structures [50]. Theorem 3.6 generalizes the main result of [57] and allows us to study the categories of multiplicative sections of such objects in a straightforward manner; as an application, we obtain the Morita invariance of such categories in §3.4 (which was proven for LA-groupoids in [80]). We also use the duality thus established to develop the Lie theory of quasi LA-groupoids, following the differentiation and integration results for LA-groupoids discussed in [23]. Moreover, we also treat L_2 -algebroids, generalizing the equivalence proven in [9] between Lie 2-algebras and 2-term L_{∞} -algebras, see §3.3.

Question (Q3) naturally leads to a subquestion:

(Q4) What is a Lie bialgebroid over a differentiable stack?

Since a Lie bialgebroid can be described as a homological vector field together with a compatible Poisson bracket, our answer to (Q2) leads naturally to an answer to (Q4) and hence to the study of *multiplicative Lie bialgebroids* §3.5. As we pointed out before, we obtain then a clear general picture for a number of constructions in [12, 24, 36, 55]. In particular, when the relevant Poisson bracket is symplectic, our multiplicative Lie bialgebroids become the cotangent groupoids of quasi Poisson groupoids §3.6.1. Cotangents of quasi Poisson groupoids are then categorified QP-manifolds of degree 1. As a result, we clarify some of the observations in [34] about the presence of a Lie 2-algebra structure on the category of multiplicative 1-forms of a quasi Poisson groupoid.

In 4 we study quasi Q-groupoids of degree two, answering the questions

- (Q5) What is a Courant algebroid over a differentiable stack?
- (Q6) What is the double of a multiplicative Lie bialgebroid?

The work in this section then promotes the ideas of [61, 84] to the setting of differentiable stacks. Thus our definition of *quasi CA-groupoids* can be regarded as a categorification of the idea of QP-manifolds of degree 2 and it includes as examples exact (twisted) CA-groupoids as in [2, 72, 79] and also the doubles of the multiplicative Lie bialgebroids described in §3.5.

In Appendix §A we include a description of double quasi Poisson groupoids which are some of the global counterparts of the multiplicative Lie bialgebroids of §3.5. In this regard, we generalize the correspondence between quasi-Manin triples of Lie 2-algebras and quasi Poisson 2-groups [35] by showing that quasi Poisson LA-groupoids over a unit manifold are indeed integrable to a double quasi Poisson groupoid under natural conditions.

1.2. Future outlook. The work presented here motivates a number of questions that mostly have to do with the global aspects of the theory. Now we shall briefly elaborate on this.

Our concept of quasi-LA groupoids should be well suited to formulate the integrability condition of quasi-Nijenhuis tensors over Lie groupoids, see [27, Remark 6.4]. In particular, one should be able to define quasi-complex structures over Lie groupoids using these objects, thus providing a suitable concept of complex structure on a differentiable stack. In the same spirit, one can use the quasi-CA groupoids that we define here as background for generalized complex structures up to homotopy. We hope to analyse these ideas in particular examples in the future.

The notions introduced here can be significantly generalized. In fact, it is natural to consider homological vector fields over graded differential higher (and derived) stacks. To do so one can work with graded simplicial manifolds and shifted multivector fields as in [81]. Notably, this would allow us to include L_{∞} -algebroids in full generality instead of their truncated versions. Also, *n*-shifted Poisson and symplectic structures should give other natural examples of these higher Lie algebroids via central extensions as in §3.6.2. However, working in such generality makes difficult to obtain explicit examples and formulae which was one of our objectives in this project.

In a different direction, one can try to find the global counterparts of quasi Q-groupoids and their morphisms. As conceptual guidance for this integration problem, one can look at three different particular cases. (1) Firstly, we have the integration of representations up to homotopy [4, 17]. Since 2-term representations up to homotopy can be viewed as an example of quasi LA-groupoid morphism, there should be a general relationship between the integration of quasi LA-groupoids morphisms and two-dimensional holonomy, as suggested by the results in [5, 17, 69]. We believe that a better understanding of this relationship would allow us to study Wilson surface observables [69] in greater generality and also in contexts more closely related to (higher) symplectic structures. (2) Secondly, it is well known that double Lie groupoids are the integrations of LA-groupoids [63] but there has been little progress integrating these objects, see [89]. We hope that combining our more flexible concept of morphism for (quasi) LA-groupoids with Sullivan's spatial realization (as applied in [88]) can shed some light on this problem. (3) Finally, we expect that integrating quasi LA-groupoids over unit groupoids should produce stacky Lie groupoids as in [98].

On a related note, one can try to quantize the categorified QP-manifolds described here. In other words, it seems interesting to study the AKSZ construction [1] but adapted to the setting of graded differentiable stacks, in the spirit of [29]. In particular, we expect that categorifying the usual Poisson sigma model should give us a way of quantizing and integrating quasi Poisson groupoids, in the same way that Poisson manifolds are quantized and integrated in [30, 31]. We hope that progress in this direction can also shed light on the classical AKSZ sigma models from a global perspective.

Acknowledgements. We thank Henrique Bursztyn, Alejandro Cabrera, Rajan Mehta, Cristian Ortiz and Chenchang Zhu for fruitful discussions and suggestions on previous drafts of this paper. D. A. thanks all the members of the Geometric Structures Laboratory at the University of Toronto for countless inspiring conversations. M.C. also thanks all the members of the Higher Structure Seminar at Göttingen.

2. QUASI Q-GROUPOIDS

In this section we introduce quasi Q-groupoids, a generalization of the Q-groupoids introduced in [72]. We approach these objects from two complementary viewpoints: one in terms

DANIEL ÁLVAREZ AND MIQUEL CUECA

of functors and natural transformations, the other one, in terms of Maurer-Cartan elements. This last description allows us to study their compatibility with multiplicative homotopy Poisson structures. We conclude this section by showing that quasi Q-groupoids form a strict 2-category.

As motivation for the general theory of graded manifolds, let us emphasize its remarkable power to express classical differential geometric ideas in a concise and conceptual manner, see §3.1 and §4.1 for the respective applications of supergeometry to Lie and Courant algebroids.

2.1. Graded manifolds. Given a non-negative integer $n \in \mathbb{N}$, let $\mathbf{V} = \bigoplus_{i=1}^{n} V_i$ be a graded vector space. An \mathbb{N} -manifold of degree n (or simply an *n*-manifold) is a ringed space $\mathcal{M} = (\mathcal{M}, C^{\bullet}(\mathcal{M}))$, where \mathcal{M} is a smooth manifold and $C^{\bullet}(\mathcal{M})$ is a sheaf of graded commutative algebras such that, for all $x \in \mathcal{M}$, there exists a neighborhood U and an isomorphism

(1)
$$C^{\bullet}(\mathcal{M})|_{U} \cong C^{\infty}(U) \otimes \operatorname{Sym} \mathbf{V}$$

of sheaves of graded commutative algebras, where Sym V denotes the graded symmetric algebra of V. The manifold M is known as the *body* of \mathcal{M} . A global section of $C^i(\mathcal{M})$ is called a *homogeneous function of degree* i; we write |f| = i for the degree of f.

A morphism of *n*-manifolds $\Psi : \mathcal{M} \to \mathcal{N}$ is a degree preserving morphism of ringed spaces, given by a pair $\Psi = (\psi, \psi^{\sharp})$, where $\psi : \mathcal{M} \to \mathcal{N}$ is a smooth map and $\psi^{\sharp} : C^{\bullet}(\mathcal{N}) \to \psi_* C^{\bullet}(\mathcal{M})$ is a degree preserving morphism of sheaves of algebras over N.

For each $n \in \mathbb{N}$, *n*-manifolds with their morphisms form a category that we denote by *n*- $\mathcal{M}an$. Now we collect some basic examples, for more details about \mathbb{N} -manifolds see e.g. [32, 70, 72, 85].

Example 2.1 (1-manifolds). Given a vector bundle $E \to M$, we can define the 1-manifold

$$E[1] = (M, \Gamma(\wedge^{\bullet} E^*)).$$

Recall that any 1-manifold is canonically of this type, see e.g. [15]. Two important examples of 1-manifolds are

$$T[1]M = (M, \Omega^{\bullet}(M))$$
 and $T^*[1]M = (M, \mathfrak{X}^{\bullet}(M)),$

where $\Omega^{\bullet}(M)$ and $\mathfrak{X}^{\bullet}(M)$ are the sheaves of differential forms and multivector fields, respectively.

Example 2.2 (Shifted tangent bundles of *n*-manifolds). One can generalize the shifted tangent bundles of ordinary manifolds considered in Example 2.1 to arbitrary *n*-manifolds $\mathcal{M} = (M, C_{\mathcal{M}})$. Rather than introducing $\Omega^{\bullet}(\mathcal{M})$ (see e.g. [70]), we will give a description using charts and coordinates as in [85].

Let U be a chart of \mathcal{M} with coordinates $\{x^{\alpha}, e^{\beta}\}$, so that $\{x^{\alpha}\}$ are coordinates on U and $C^{\bullet}(\mathcal{M})|_{U} = C^{\infty}(U) \otimes \text{Sym} \mathbf{V}$ with $\mathbf{V} = \langle e^{\beta} \rangle$. Just as for ordinary smooth manifolds, we can introduce the differentials of the coordinates, $\{dx^{\alpha}, de^{\beta}\}$, with the transformation rule

$$\begin{cases} \widehat{x}^{\alpha} = F^{\alpha}(x), \\ \widehat{e}^{\beta} = F^{\beta}(x,e) \end{cases} \implies \begin{cases} d\widehat{x}^{\alpha} = \frac{\partial F^{\alpha}(x)}{\partial x^{\alpha'}} dx^{\alpha'}, \\ d\widehat{e}^{\beta} = \frac{\partial F^{\beta}(x,e)}{\partial x^{\alpha}} dx^{\alpha} + \frac{\partial F^{\beta}(x,e)}{\partial e^{\beta'}} dx^{\beta'}. \end{cases}$$

For $k \ge 1$ we define the (n+k)-manifold $T[k]\mathcal{M} = (M, C_{T[k]\mathcal{M}})$ by

$$C^{\bullet}(T[k]\mathcal{M})|_{U} = C^{\infty}(U) \otimes \operatorname{Sym} \mathbf{W} \quad \text{with} \quad \mathbf{W} = \mathbf{V} \oplus \langle dx^{\alpha}, de^{\beta} ; \ |dx^{\alpha}| = k, \ |de^{\beta}| = |e^{\beta}| + k \rangle.$$

Example 2.3 (Vector and multivector fields). Let \mathcal{M} be an *n*-manifold. A vector field of degree k on \mathcal{M} is a degree k derivation X of the graded algebra $C^{\bullet}(\mathcal{M})$, i.e., an \mathbb{R} -linear map X:

 $C^{\bullet}(\mathcal{M}) \to C^{\bullet+k}(\mathcal{M})$ with the property that, for all $f, g \in C_{\mathcal{M}}$ with f homogeneous, |X(f)| = |f| + k and

(2)
$$X(fg) = X(f)g + (-1)^{|f|k} f X(g).$$

Vector fields give rise to a sheaf of $C^{\bullet}(\mathcal{M})$ -modules over M. The sheaf of degree k vector fields is denoted by $\mathfrak{X}^{1,k}(\mathcal{M})$. The graded commutator of vector fields, defined for homogeneous vector fields X, Y by

$$[X,Y] = XY - (-1)^{|X||Y|}YX$$

makes $\mathfrak{X}^{1,\bullet}(\mathcal{M})$ into a sheaf of graded Lie algebras. A degree 1 vector field on an *n*-manifold $\mathcal{M}, Q \in \mathfrak{X}^{1,1}(\mathcal{M})$, such that

$$[Q,Q] = 2Q^2 = 0$$

is called a *homological vector field*. Note that, in this case, $(C^{\bullet}(\mathcal{M}), Q)$ is a differential complex, justifying the terminology. The pair (\mathcal{M}, Q) will be referred as a *Q*-manifold.

Till now we just defined vector fields on an *n*-manifold \mathcal{M} . However, we also need to introduce *multivector fields*. They are the elements of the graded symmetric algebra

$$\left(\bigoplus_{i} \operatorname{Sym}^{i} \mathfrak{X}^{1,\bullet}(\mathcal{M}) = \mathfrak{X}^{i,\bullet}(\mathcal{M}), \wedge\right) \quad \text{with} \quad X \wedge Y = (-1)^{1 \cdot 1 + |X||Y|} Y \wedge X \quad X, Y \in \mathfrak{X}^{1,\bullet}(\mathcal{M}).$$

If we extend the Lie bracket using the Leibniz rule, this algebra carries a Poisson bracket of bi-degree (-1, 0) that we denote by $[\cdot, \cdot]$ and it is known as the *Schouten bracket*.

2.2. Lie groupoids and algebroids in the category of graded manifolds. Following [70, 72], see also [21], we say that an *n*-graded Lie groupoid¹ is a Lie groupoid object in the category *n*- $\mathcal{M}an$ such that its source and target maps are submersions. More concretely, this means a collection of data ($\mathcal{G} \rightrightarrows \mathcal{M}, \mathbf{s}, \mathbf{t}, \mathbf{m}, \mathbf{u}, \mathbf{i}$) where $\mathcal{G} = (G, C^{\bullet}(\mathcal{G}))$ and $\mathcal{M} = (M, C^{\bullet}(\mathcal{M}))$ are two *n*-manifolds and $\mathbf{s}, \mathbf{t} : \mathcal{G} \rightarrow \mathcal{M}, \mathbf{m} : \mathcal{G}^{(2)} = \mathcal{G}_{\mathbf{s}} \times_{\mathbf{t}} \mathcal{G} \rightarrow \mathcal{G}, \mathbf{u} : \mathcal{M} \rightarrow \mathcal{G}$ and $\mathbf{i} : \mathcal{G} \rightarrow \mathcal{G}$ are graded morphisms that make the following diagrams commute

It follows from the definition that the underlying body $G \rightrightarrows M$ of an *n*-graded groupoid is a usual Lie groupoid.

An *n*-graded Lie groupoid morphism between $\mathcal{G} \rightrightarrows \mathcal{M}$ and $\mathcal{H} \rightrightarrows \mathcal{N}$ is a pair of *n*-manifold morphisms $\Phi_1 : \mathcal{G} \rightarrow \mathcal{H}$ and $\Phi_0 : \mathcal{M} \rightarrow \mathcal{N}$ for which the following diagrams commute

\mathcal{G} —	$\xrightarrow{P_1} \mathcal{H}$		$\mathcal{G}^{(2)}$ –	$\xrightarrow{\mathcal{H}^{(2)}} \mathcal{H}^{(2)}$
st	st	and	m _𝔅	m _H
$\mathcal{M}_{-\frac{1}{4}}$	$_{0} \mathcal{N}$		\mathcal{G} —	${\mathbb{P}_1} \mathcal{H}.$

Example 2.4 (Shifted tangent groupoids). Let $(\mathcal{G} \Rightarrow \mathcal{M}, \mathbf{s}, \mathbf{t}, \mathbf{m}, \mathbf{u}, \mathbf{i})$ be an *n*-graded Lie groupoid. Then for any $j \in \mathbb{N}$ we have that $(T[j]\mathcal{G} \Rightarrow T[j]\mathcal{M}, T[j]\mathbf{s}, T[j]\mathbf{t}, T[j]\mathbf{m}, T[j]\mathbf{u}, T[j]\mathbf{i})$ defines an (n+j)-graded Lie groupoid.

¹Do not confuse with the more familiar notion of Lie n-groupoid.

Example 2.5 (Morita equivalence for graded Lie groupoids). As in the classical case, one can declare two *n*-graded Lie groupoids \mathcal{G} and \mathcal{H} to be Morita equivalent if there is a zig-zag of weak equivalences $\mathcal{G} \leftarrow \mathcal{K} \rightarrow \mathcal{H}$. The definition of a weak equivalence is exactly the same as the classical one given in [77, §5.4].

Analogously to the classical theory, we have that an *n*-graded Lie algebroid is a Lie algebroid object in the category *n*- $\mathcal{M}an$, see [70, 71]. More concretely, such a structure consists of a collection of data $(\mathcal{A} \to \mathcal{M}, \rho, [\cdot, \cdot])$ where \mathcal{A} and \mathcal{M} are two *n*-manifolds that forms a vector bundle $\mathcal{A} \to \mathcal{M}, \rho : \mathcal{A} \to T\mathcal{M}$ is a vector bundle morphism covering the identity and $[\cdot, \cdot]$: $\Gamma^i \mathcal{A} \times \Gamma^j \mathcal{A} \to \Gamma^{i+j} \mathcal{A}$ is a bracket, where $\Gamma^i \mathcal{A}$ denotes the degree *i* sections of $\mathcal{A} \to \mathcal{M}$, satisfying:

- (1) $[a_1, a_2] = -(-1)^{|a_1||a_2|} [a_2, a_1],$
- (2) $[a_1, fa_2] = \rho(a_1)(f)a_2 + (-1)^{|a_1||f|}f[a_1, a_2],$
- (3) $[a_1, [a_2, a_3]] = [[a_1, a_2], a_3] + (-1)^{|a_1||a_2|} [a_2, [a_1, a_3]],$

for $a_1, a_2, a_3 \in \Gamma \mathcal{A}, f \in C_{\mathcal{M}}$.

The construction of the *n*-graded Lie algebroid of an *n*-graded Lie groupoid is identical to the classical one [70, §3.2.1]. Let $\mathcal{G} \rightrightarrows \mathcal{M}$ be an *n*-graded Lie groupoid, then the vector bundle $\mathcal{A} = \mathbf{u}^* \ker(T\mathbf{s}) \rightarrow \mathcal{M}$ has the structure of a graded algebroid where the anchor is given by $\rho = T\mathbf{t}$ and the bracket is given by the identification

 $\Gamma \mathcal{A} \cong \{ \text{ Right invariant vector fields on } \mathcal{G} \}$

where a right (resp. left) invariant vector field $X \in \mathfrak{X}^{1,j}(\mathcal{G})$ is defined by the equations

(3)
$$T[j]\mathbf{s} \circ X = 0 \text{ and } T[j]\mathbf{m} \circ (X \times 0) \circ ((\mathbf{u} \circ \mathbf{t}) \times \mathrm{Id}) \circ \Delta = X$$

(4)
$$(resp. T[j]\mathbf{t} \circ X = 0 \text{ and } T[j]\mathbf{m} \circ (0 \times X) \circ (\mathrm{Id} \times (\mathbf{u} \circ \mathbf{s})) \circ \Delta = X),$$

with $0: \mathcal{G} \to T[j]\mathcal{G}$ the zero section and $\Delta: \mathcal{G} \to \mathcal{G} \times \mathcal{G}$ the diagonal map.

Example 2.6. (The shifted tangent algebroids) Let $\mathcal{G} \rightrightarrows \mathcal{M}$ be an *n*-graded Lie groupoid with *n*-graded Lie algebroid $\mathcal{A} \rightarrow \mathcal{M}$ then we have that $T[k]\mathcal{A} \rightarrow \mathcal{M}$ are (n+k)-graded Lie algebroids. In fact, $\mathcal{A}_{T[k]\mathcal{G}} \cong T[k]\mathcal{A}$. Moreover, we can see that $T[k]\mathcal{G} \rightarrow T[k]\mathcal{M}$ is a VB-groupoid with base groupoid $\mathcal{G} \rightarrow \mathcal{M}$ and core given by $\mathcal{A}[k] \rightarrow \mathcal{M}$. A similar description holds for the corresponding (n+k)-graded Lie algebroids, see [66, §11.2] for the classical picture.

2.3. Homological vector fields on graded Lie groupoids. The next step is to study vector fields on *n*-graded Lie groupoids. Let $\mathcal{G} \rightrightarrows \mathcal{M}$ be an *n*-graded Lie groupoid with *n*-graded Lie algebroid $\mathcal{A} \rightarrow \mathcal{M}$. There are three classes of distinguished vector fields *left invariant*, *right invariat* and *multiplicative*.

As we already pointed out, left and right invariant vector fields can be identified with sections of the *n*-graded Lie algebroid. More explicitly, if we take a section of degree $j, a \in \Gamma^{j} \mathcal{A}$, then we can define its corresponding right and left invariant vector fields on \mathcal{G} by the formulas

$$a^r = T[j] \mathtt{m} \circ (a \circ \mathtt{t} imes 0) \circ \Delta \quad ext{ and } \quad a^l = T[j] \mathtt{m} \circ (0 imes a \circ \mathtt{s}) \circ \Delta.$$

We say that $X \in \mathfrak{X}^{1,k}(\mathcal{G})$ is a multiplicative vector field if there is $X_0 \in \mathfrak{X}^{1,k}(\mathcal{M})$ such that



is a groupoid morphism. Notice that for any $a \in \Gamma^{j} \mathcal{A}$, the vector field $X = a^{l} - a^{r}$ is multiplicative with $X_{0} = \rho(a)$. We denote the multiplicative vector fields by $\mathfrak{X}_{mul}^{1,\bullet}(\mathcal{G})$

Remark 2.7. Our definition of multiplicative vector fields follows the classical definition introduced in [67, $\S3.3$]. In [72, $\S3.7$] the simplicial approach is adopted. It is straightforward to see that both definitions coincide.

Definition 2.8. An *n*-graded quasi Q-groupoid $(\mathcal{G} \Rightarrow \mathcal{M}, Q, q)$ is an *n*-graded Lie groupoid $\mathcal{G} \Rightarrow \mathcal{M}$ endowed with a degree 1 multiplicative vector field $Q \in \mathfrak{X}^{1,1}_{mul}(\mathcal{G})$ and a degree 2 section of the graded algebroid $q \in \Gamma^2 \mathcal{A}$ satisfying

$$Q^2 = q^l - q^r \qquad \text{and} \qquad [Q, q^r] = 0.$$

When q = 0 we recover the Q-groupoids introduced in [70, 72]. We postpone the discussion of non-trivial examples to §3 and §4.

Multiplicative vector fields have an infinitesimal counterpart given by IM^2 vector fields. To be precise, we say that $X \in \mathfrak{X}^{1,k}(\mathcal{A})$ is an *IM vector field* if there is $X_0 \in \mathfrak{X}^{1,k}(\mathcal{M})$ such that

$$\mathcal{A} \xrightarrow{X} T[k]\mathcal{A}$$

$$\downarrow \qquad \qquad \downarrow$$

$$\mathcal{M} \xrightarrow{X_0} T[k]\mathcal{M}$$

gives a graded Lie algebroid morphism.

Definition 2.9. An *n*-graded quasi Q-algebroid $(\mathcal{A} \to \mathcal{M}, Q, q)$ is an *n*-graded Lie algebroid $\mathcal{A} \to \mathcal{M}$ endowed with a degree 1 IM vector field $Q \in \mathfrak{X}^{1,1}(\mathcal{A})$ and a degree 2 section $q \in \Gamma^2 \mathcal{A}$ satisfying

$$Q^2 = [q, \cdot]$$
 and $Q(q) = 0.$

When q = 0 we recover the Q-algebroids introduced in [70, 71]. If in addition we have $([\cdot, \cdot] = 0, \rho = 0)$ we get the Q-bundles of [53].

Thus, one obtains the following natural differentiation result.

Proposition 2.10. Let $(\mathcal{G} \rightrightarrows \mathcal{M}, Q, q)$ be an n-graded quasi Q-groupoid then $(\mathcal{A} \rightarrow \mathcal{M}, \text{Lie}(Q), q)$ is an n-graded quasi Q-algebroid.

In order to shed light on the definition of quasi Q-structures we re-interpret them in several ways. In §2.3.1 we show how q gives a natural transformation between Q^2 and 0. In §2.3.2 we demonstrate that quasi Q-structures are Maurer-Cartan elements. Finally, in Proposition 3.35 we characterize them as actions of $\mathbb{R}[-1]$ on n-graded Lie groupoids.

2.3.1. Interpretation in terms of natural transformations. Here we describe the $q \in \Gamma^2 \mathcal{A}$ as a natural transformation between Q^2 and the zero vector field. In this way, we generalize the work of [9], where the Jacobiator of a Lie 2-algebra is defined as a natural transformation. We explain this relation in §3.3.

Let $(\Phi_1, \Phi_0), (\Psi_1, \Psi_0) : (\mathcal{G} \rightrightarrows \mathcal{M}) \rightarrow (\mathcal{H} \rightrightarrows \mathcal{N})$ be two graded groupoid morphisms. Recall that a *natural transformation* $\alpha : (\Phi_1, \Phi_0) \Rightarrow (\Psi_1, \Psi_0)$ is a graded manifold morphism $\alpha : \mathcal{M} \rightarrow \mathcal{H}$ satisfying the following equations

(5)
$$\mathbf{s}_{\mathcal{H}} \circ \alpha = \Phi_0$$
, $\mathbf{t}_{\mathcal{H}} \circ \alpha = \Psi_0$ and $\mathbf{m}_{\mathcal{H}} \circ (\Phi_1 \times \alpha) \circ (\mathrm{Id} \times \mathbf{s}_{\mathcal{G}}) = \mathbf{m}_{\mathcal{H}} \circ (\alpha \times \Psi_1) \circ (\mathbf{t}_{\mathcal{G}} \times \mathrm{Id})$.

²IM stands from Inifnitesimally Multiplicative.

Proposition 2.11. Let $\mathcal{G} \rightrightarrows \mathcal{M}$ be a graded Lie groupoid and $Q \in \mathfrak{X}^{1,1}_{mul}(\mathcal{G})$. Then $q \in \Gamma^2 \mathcal{A}$ satisfying $Q^2 = q^l - q^r$ is equivalent to a natural transformation



such that $\pi_{\mathcal{G}} \circ \alpha = \mathfrak{u}$, where $\pi_{\mathcal{G}} : T[2]\mathcal{G} \to \mathcal{G}$ is the projection. Moreover $[Q, q^r] = 0$ if and only if α preserves Q, i.e. $T[2]Q \circ \alpha = T[1]\alpha \circ Q_0$.

Proof. Let α be a natural transformation satisfying $\pi_{\mathcal{G}} \circ \alpha = \mathfrak{u}$. Then $T[2]\mathfrak{s} \circ \alpha = 0_{\mathcal{M}}$ and so, combining this fact with equation $\pi_{\mathcal{G}} \circ \alpha = \mathfrak{u}$, we get that α is a degree 2 section of \mathcal{A} . Since α is a natural transformation we have that the diagram



is commutative. Thus $Q^2 = T[2]\mathfrak{m}((T[2]\mathfrak{i}) \circ \alpha \circ \mathfrak{t} + 0_{\mathcal{G}}, \alpha \circ \mathfrak{s})$ and hence

(7)
$$Q^2 = T[2]\mathfrak{m}((T[2]\mathfrak{i}) \circ \alpha \circ \mathfrak{t} + 0_{\mathcal{G}}, 0_{\mathcal{G}} + \alpha \circ \mathfrak{s}) = \alpha^l + ((T[2]\mathfrak{i}) \circ \alpha)^r = \alpha^l - \alpha^r,$$

since T[2]m is linear on pairs of composable sections. Conversely, starting from $q \in \Gamma^2 \mathcal{A}$ satisfying $Q^2 = q^l - q^r$ we define $\alpha : \mathcal{M} \to T[2]\mathcal{G}$ as $q : \mathcal{M} \to \mathcal{A}[2] \subset T[2]\mathcal{G}$. Then we get that

$$T[2]\mathbf{s}(\alpha) = T[2]\mathbf{s}(q) = 0, \quad T[2]\mathbf{t}(\alpha) = T[2]\mathbf{t}(q) = \rho(q) = Q[2]\mathbf{t}(q) = 0$$

and by (7) the diagram (6) commutes and so α is a natural transformation from $0_{\mathcal{G}}$ to Q^2 .

For the moreover part, notice that we can view Q and α as derivations. Thus, we get that condition $[Q, q^r] = 0$ is equivalent to the fact that α preserves Q.

2.3.2. Interpretation in terms of the Maurer-Cartan equation. Let $G \rightrightarrows M$ be a Lie groupoid with Lie algebroid $(A \rightarrow M, \rho, [\cdot, \cdot])$. It is shown in [80, Example 7.2] that $\Gamma A \rightarrow \mathfrak{X}_{mul}(G)$ inherits a dgla³ structure given by

$$d(a) = a^r - a^l$$
, $[X, Y] = [X, Y]$, $[X, a] = [X, a^r] \circ \mathbf{u}$ and $[a, b] = 0$,

where $a, b \in \Gamma A$, $X, Y \in \mathfrak{X}_{mul}(G)$. Moreover, in [80, Prop 8.1] they show that if two groupoids are Morita equivalent, then the associated dgla's are L_{∞} quasi-isomorphic. As explained in [46], the above dgla codifies the vector fields on the differentiable stack presented by $G \rightrightarrows M$. For *n*-graded Lie groupoids we have an analogous result.

³Differential graded Lie algebra.

Proposition 2.12. Let $\mathcal{G} \rightrightarrows \mathcal{M}$ be an n-graded Lie groupoid and $\mathcal{A} \rightarrow \mathcal{M}$ its n-graded Lie algebroid. Define $L_{\mathcal{G}} = \bigoplus_{i \in \mathbb{Z}} L^i$ with $L^i = \Gamma^{i+1} \mathcal{A} \oplus \mathfrak{X}^{1,i}_{mul}(\mathcal{G})$. Then $L_{\mathcal{G}}$ inherits a dgla structure given by

$$d(a) = a^r - a^l$$
, $[X, Y] = [X, Y]$, $[X, a] = [X, a^r] \circ u$ and $[a, b] = 0$,

where $a, b \in \Gamma^{\bullet} \mathcal{A}, X, Y \in \mathfrak{X}_{mul}^{1,\bullet}(\mathcal{G})$. Moreover, if two n-graded Lie groupoid are Morita equivalent then the associated dgla's are L_{∞} quasi-isomorphic.

Proof. This is analogous to the classical case presented in [80].

Proposition 2.13. Let $\mathcal{G} \rightrightarrows \mathcal{M}$ be an n-graded Lie groupoid. Then there is a canonical one to one correspondence between:

- Maurer-Cartan elements of $(L_{\mathcal{G}}, d, [\cdot, \cdot])$.
- Quasi Q-groupoid structures on $\mathcal{G} \rightrightarrows \mathcal{M}$.

Proof. Recall that a *Maurer-Cartan element* on a dgla is a degree 1 element X satisfying $dX + \frac{1}{2}[X, X] = 0$. In our case, $L^1_{\mathcal{G}} = \Gamma^2 \mathcal{A} \oplus \mathfrak{X}^{1,1}_{mul}(\mathcal{G})$ then X = q + Q and we have that the Maurer-Cartan equation decomposes into

$$0 = q^r - q^l + \frac{1}{2}[Q, Q] = q^r - q^l + Q^2$$
 and $0 = [Q, q] = [Q, q^r] \circ \mathbf{u}$

which are exactly the equations defining a quasi Q-groupoid structure. The other direction is analogous.

Remark 2.14 (Gauge transformations). The degree 0 elements on a dgla act by gauge transformations on Maurer-Cartan elements, see e.g. [54, §5.1]. In our case, $L^0_{\mathcal{G}} = \Gamma^1 \mathcal{A} \oplus \mathfrak{X}^{1,0}_{mul}(\mathcal{G})$ so the gauge transformation of $X \in \mathfrak{X}^{1,0}_{mul}(\mathcal{G})$ is just a usual transformation by a groupoid diffeomorphism and therefore expected. But for a $b \in \Gamma^1 \mathcal{A}$ we get the gauge transformation

$$e^{b} \cdot (q,Q) = (q + [Q,b^{r}] \circ \mathbf{u} - \frac{1}{2}[b,b], \ Q + b^{l} - b^{r}) \text{ for } (q,Q) \in L^{1}_{\mathcal{G}}.$$

In [20], Q-algebroids with $Q = [b, \cdot]$ for $b \in \Gamma^1 \mathcal{A}$ where studied. In our language, those are the gauge equivalent to zero with [b, b] = 0.

Some immediate and interesting consequences of the above propositions are the following.

Corollary 2.15. Let $\mathcal{G} \rightrightarrows \mathcal{M}$ be an n-graded Lie groupoid and (Q,q) a quasi Q-groupoid structure. Small deformations of the quasi Q-groupoid structure (Q,q) are controlled by the twisted dgla $(L_{\mathcal{G}}, d + [Q+q, \cdot], [\cdot, \cdot])$.

For more detail on L_{∞} -algebras and deformations see [54] and references therein.

Corollary 2.16. *Quasi Q-groupoid structures can be transported along Morita equivalences of graded Lie groupoids.*

Proof. This follows from the fact that L_{∞} quasi-isomorphisms between strict Lie 2-algebras induce bijections between moduli sets of Maurer-Cartan elements, see [16, Corollary A.16].

Observe that the concept of Q-groupoid is not Morita invariant. More concretely, let $(\mathcal{G} \Rightarrow \mathcal{M}, Q)$ be an *n*-graded Q-groupoid and $\mathcal{H} \Rightarrow \mathcal{N}$ another *n*-graded Lie groupoid Morita equivalent to $\mathcal{G} \Rightarrow \mathcal{M}$. Thus $\mathcal{H} \Rightarrow \mathcal{N}$ inherits naturally a quasi Q-groupoid structure by the above Corollary but, in general, this is not automatically a Q-groupoid structure.

2.4. Quasi QP-groupoids. On a Lie groupoid $G \Rightarrow M$, the dgla $\Gamma A \to \mathfrak{X}_{mul}(G)$ described in §2.3.2 can be extended to include multivector fields as explained in [16]. The underlying graded vector space is given by $V_G^{i-1} = \Gamma \wedge^{i+1} A \oplus \mathfrak{X}_{mul}^i(G)$ with differential and bracket

(8) $d(\pi + \Pi) = 0 \oplus \pi^r - \pi^l$, $[\pi_1 \oplus \Pi_1, \pi_2 \oplus \Pi_2] = (-1)^k \Pi_1 \cdot \pi_2 - (-1)^{(k+1)l} \Pi_2 \cdot \pi_1 \oplus [\Pi_1, \Pi_2]$

where $\pi_1 \oplus \Pi_1 \in V_G^k$, $\pi_2 \oplus \Pi_2 \in V_G^l$ and $\Pi_1 \cdot \pi_2 \in \Gamma \wedge^{k+l+2} A$ denotes the unique section such that $[\Pi_1, \pi_2^r] = (\Pi_1 \cdot \pi_2)^r$. The importance of this dgla comes from the fact that its Maurer-Cartan elements are exactly quasi Poisson structures, i.e. $\pi + \Pi \in V_G^1 = \Gamma \wedge^3 A \oplus \mathfrak{X}_{mul}^2(G)$ such that

(9)
$$\pi^{l} - \pi^{r} = \frac{1}{2}[\Pi, \Pi], \text{ and } [\Pi, \pi^{r}] = 0$$

This dgla extends to the graded context as we show now. Let $\mathcal{G} \rightrightarrows \mathcal{M}$ be an *n*-graded Lie groupoid. Define

$$\mathcal{V}_{\mathcal{G}}^{i-(n+1)} = igoplus_{(n+1)j+k=i} \Gamma^{k+1} \wedge^{j} \mathcal{A} \oplus \mathfrak{X}_{mul}^{j,k}(\mathcal{G})$$

with differential and bracket given by the same formulas as before. It is easy to see that

$$\mathcal{V}_{\mathcal{G}}^{1} = C^{n+3}(\mathcal{M}) \oplus C^{n+2}_{mul}(\mathcal{G}) \oplus \Gamma^{2}\mathcal{A} \oplus \mathfrak{X}^{1,1}_{mul}(\mathcal{G}) \oplus \Gamma^{1-n} \wedge^{2}\mathcal{A} \oplus \mathfrak{X}^{2,-n}_{mul}(\mathcal{G}) \oplus \Gamma^{-2n} \wedge^{3}\mathcal{A} \cdots$$

By inspection one can see that its Maurer-Cartan elements include quasi Q-structures as well as degree -n quasi Poisson structures. Moreover, we have a natural notion of compatibility for two such structures.

Definition 2.17. Let $\mathcal{G} \rightrightarrows \mathcal{M}$ be an *n*-graded Lie groupoid. A quasi *QP*-structure is a $q \in \Gamma^2 \mathcal{A}, \ Q \in \mathfrak{X}^{1,1}_{mul}(\mathcal{G}), \ \Pi \in \mathfrak{X}^{2,-n}_{mul}(\mathcal{G}) \text{ and } \pi \in \Gamma^{-2n} \wedge^3 \mathcal{A} \text{ satisfying}$

$$Q^2 = q^l - q^r$$
, $\frac{1}{2}[\Pi, \Pi] = \pi^l - \pi^r$, and $[Q, q^r] = [\Pi, \pi^r] = [Q, \Pi] = [Q, \pi^r] = [\Pi, q^r] = 0$.

When Q = 0 and q = 0 we say that $(\mathcal{G} \rightrightarrows \mathcal{M}, \Pi, \pi)$ is a *degree n* quasi *P*-groupoid and when q = 0 and $\pi = 0$ then $(\mathcal{G} \rightrightarrows \mathcal{M}, Q, \Pi)$ is a degree *n QP*-groupoid as studied in [60].

Remark 2.18. A general Maurer-Cartan element of the dgla $(\mathcal{V}_{\mathcal{G}}^{\bullet}, d, [\cdot, \cdot])$ is a multiplicative analogue of the homotopy Poisson structures studied in [73]. In §3.5 we study the degree 1 case. For other strict versions of Poisson structure on graded groupoids see [21].

A particularly relevant case for us is given by assuming that $\pi = 0$ and Π is a non-degenerate bivector field, i.e. that Π is the inverse of a symplectic structure. In this case, the vector field Q is always hamiltonian, i.e. $Q = \{\Theta, \cdot\}$ for some $\Theta \in C_{mul}^{n+1}(\mathcal{G})$, see [85]. Therefore one can also impose the following condition on q.

Definition 2.19. A degree n symplectic quasi Q-groupoid $(\mathcal{G} \Rightarrow \mathcal{M}, \Pi, \Theta, \theta)$ is an n-graded Lie groupoid $\mathcal{G} \Rightarrow \mathcal{M}$ equipped with $\Pi \in \mathfrak{X}^{2,-n}_{mul}(\mathcal{G})$ which is Poisson and non-degenerate, a $\Theta \in C^{n+1}_{mul}(\mathcal{G})$ and a $\theta \in C^{n+2}(\mathcal{M})$ satisfying

(10)
$$\frac{1}{2}\{\Theta,\Theta\} = \delta\theta = \mathbf{s}^*\theta - \mathbf{t}^*\theta \quad \text{and} \quad \{\Theta,\mathbf{t}^*\theta\} = 0;$$

where $\{\cdot, \cdot\}$ is the Poisson bracket determined by Π .

Corollary 2.20. A degree n symplectic quasi Q-groupoid ($\mathcal{G} \rightrightarrows \mathcal{M}, \Pi, \Theta, \theta$) defines a quasi QP-groupoid with

$$Q = \{\Theta, \cdot\} \quad and \quad q^r = \{\mathsf{t}^*\theta, \cdot\}.$$

Proof. Follows directly from the fact that Hamiltonian vector fields preserves the Poisson structure together with $X_{\{f,g\}} = [X_f, X_g]$ for $f, g \in C(\mathcal{G})$.

Proposition 2.21. Let $\mathcal{G} \rightrightarrows \mathcal{M}$ be an n-graded Lie groupoid with n-graded Lie algebroid \mathcal{A} .

- (1) For all l > n, $(T^*[l]\mathcal{G} \rightrightarrows \mathcal{A}^*[l], \Pi_{can})$ is a degree l symplectic groupoid.
- (2) There is a one to one correspondence between:

 - Maurer-Cartan elements in (V[•]_G, d, [·, ·]).
 Pairs Θ ∈ Cⁿ⁺²_{mul}(T*[n+1]G) and θ ∈ Cⁿ⁺³(A*[n+1]) satisfying (10).

Proof. The first statement is a graded analog of the classical result and the proof is identical, see e.g. $[66, \S11.3]$. For the second statement, notice that

$$\mathcal{V}_{\mathcal{G}}^{1} = \bigoplus_{(n+1)j+k=n+2} \Gamma^{k+1} \wedge^{j} \mathcal{A} \oplus \mathfrak{X}_{mul}^{j,k}(\mathcal{G})$$
$$= \left(\bigoplus_{(n+1)j+k=n+3} \Gamma^{k} \wedge^{j} \mathcal{A}\right) \oplus \left(\bigoplus_{(n+1)j+k=n+2} \mathfrak{X}_{mul}^{j,k}(\mathcal{G})\right)$$
$$= C^{n+3}(\mathcal{A}^{*}[n+1]) \oplus C_{mul}^{n+2}(T^{*}[n+1]\mathcal{G}).$$

Thus, elements in \mathcal{V}^1 are identified with pairs (θ, Θ) . Under this identification, the Maurer-Cartan equation

$$d(\theta, \Theta) + \frac{1}{2}[(\theta, \Theta), (\theta, \Theta)] = 0$$

10) because $d\theta = \theta^r - \theta^l = \mathbf{t}^* \theta - \mathbf{s}^* \theta.$

decomposes into the equations (10) because $d\theta = \theta^r - \theta^l = t^*\theta - s^*\theta$.

2.5. Morphisms of quasi Q-groupoids. Here we show that quasi Q-groupoids naturally form a strict 2-category. We will show in §3.3 that this strict 2-category extends the category of semistrict Lie 2-algebras [9].

2.5.1. 1-morphisms. Given two quasi Q-groupoids $(\mathcal{G} \rightrightarrows \mathcal{M}, Q_{\mathcal{G}}, q_{\mathcal{G}})$ and $(\mathcal{H} \rightrightarrows \mathcal{N}, Q_{\mathcal{H}}, q_{\mathcal{H}})$, a morphism (or 1-morphism) between them is a pair (F, τ) consisting of a morphism of graded Lie groupoids $F: \mathcal{G} \to \mathcal{H}$ together with a natural transformation $\tau: T[1]F \circ Q_{\mathcal{G}} \Rightarrow Q_{\mathcal{H}} \circ F$ as in the following diagram



such that $\pi_{F^*T[1]\mathcal{H}} \circ \tau = \mathfrak{u}_{\mathcal{G}}$ and the following coherence law is satisfied: the two natural transformations $0 \Rightarrow Q_{\mathcal{H}}^2 \circ F$ we can define canonically in this situation coincide, i.e.

(11)
$$q_{\mathcal{H}} \diamond 1_F = \left(1_{T[1]Q_{\mathcal{H}}} \diamond \tau\right) \bullet \left(T[1]\tau \diamond 1_{Q_{\mathcal{G}}}\right) \bullet \left(1_{T[2]F} \diamond q_{\mathcal{G}}\right);$$

where \diamond and \bullet denote, respectively, the horizontal and vertical compositions of natural transformations.

To make sense of (11) notice that $0 = T[2]F \circ 0_{\mathcal{G}} = 0_{\mathcal{H}} \circ F$ and that for a vector field $Q \in \mathfrak{X}^{1,k}(\mathcal{M})$ we get that $T[1]Q: T[1]\mathcal{M} \to T[1]T[1]\mathcal{M}$ identifies $T[1]Q = \mathcal{L}_Q$ with the Lie derivative because $C_{T[1]\mathcal{M}} \cong \Omega(\mathcal{M})$. However, $T[1]T[1]\mathcal{M}$ is a double vector bundle with core $T[2]\mathcal{M}$ and we get $Q^2 = T[1]Q \circ Q$ restricted to the core.

With these identifications in mind, the left hand side in (11) is the natural transformation



while the right hand side in (11) is the composition of the natural transformations



We will see later several situations that are clarified because of the introduction of this concept. Let us start by relating 1-morphisms with gauge transformations.

Proposition 2.22. Let $(1, \tau) : (\mathcal{G} \rightrightarrows \mathcal{M}, Q, q) \rightarrow (\mathcal{G} \rightrightarrows \mathcal{M}, Q', q')$ be a 1-morphism. Then τ is given by $b \in \Gamma^1 \mathcal{A}$ and the quasi Q-structures are gauge equivalent, i.e. $e^b \cdot (q, Q) = (q', Q')$.

Proof. By definition, a 1-morphism $(1, \tau)$ is given by a natural transformation $\tau : Q \Rightarrow Q'$ that we can also view as a natural transformation $\tau : 0_{\mathcal{G}} \Rightarrow Q' - Q$. So it is determined by a map $\tau : \mathcal{M} \to T[1]\mathcal{G}$ satisfying $T[1]\mathbf{s}_{\mathcal{G}} \circ \tau = 0$ and $\pi_{T[1]\mathcal{G}} \circ \tau = \mathbf{u}_{\mathcal{G}}$ so τ is given by a degree 1 section of the graded Lie algebroid $b \in \Gamma^1 \mathcal{A}$. Moreover, the last equation in (5) implies that the following diagram is commutative



Thus, using the same trick as in (6), we get $Q' - Q = b^l - b^r$. Since (q, Q) and (q', Q') are quasi Q-groupoids, we get

(12)
$$(q')^{l} - (q')^{r} = Q' \circ Q' = Q' \circ (b^{l} - b^{r} + Q) = Q' \circ (b^{l} - b^{r}) + (b^{l} - b^{r} + Q) \circ Q$$
$$= Q' \circ (b^{l} - b^{r}) + (b^{l} - b^{r}) \circ Q + q^{l} - q^{r}.$$

Since F = Id, the horizontal multiplication \diamond of natural transformations corresponds to composition of vector fields while the vertical multiplication \bullet is just the sum of vector fields. These observations imply that (11) is equivalent to (12). Finally, observe that if we apply $Q' = b^l - b^r + Q$ in the last term of (12) we get the equation

$$\begin{aligned} (q')^{l} - (q')^{r} &= (b^{l} - b^{r} + Q) \circ (b^{l} - b^{r}) + (b^{l} - b^{r}) \circ Q + q^{l} - q^{r} \\ &= -\frac{1}{2}[b,b]^{l} - [b^{r},b^{l}] + \frac{1}{2}[b,b]^{r} + [Q,b^{l}] - [Q,b^{r}] + q^{l} - q^{r}; \\ &= (q',Q') = e^{b} \cdot (q,Q) \text{ as we want.} \end{aligned}$$

meaning that $(q', Q') = e^b \cdot (q, Q)$ as we want.

2.5.2. Composition of 1-morphisms. Let (F_1, τ_1) : $(\mathcal{G}, Q_{\mathcal{G}}, q_{\mathcal{G}}) \rightarrow (\mathcal{H}, Q_{\mathcal{H}}, q_{\mathcal{H}})$ and (F_2, τ_2) : $(\mathcal{H}, Q_{\mathcal{H}}, q_{\mathcal{H}}) \to (\mathcal{K}, Q_{\mathcal{K}}, q_{\mathcal{K}})$ be morphisms of quasi Q-groupoids. Then their composition is $(F_2 \circ F_1, (\tau_2 \diamond 1_{F_1}) \bullet (1_{T[1|F_1} \diamond \tau_1)),$ according to the diagram below



2.5.3. 2-morphisms. Finally, given two 1-morphisms $(F, \tau), (F', \tau') : (\mathcal{G}, Q_{\mathcal{G}}, q_{\mathcal{G}}) \to (\mathcal{H}, Q_{\mathcal{H}}, q_{\mathcal{H}})$ a 2-morphism between them is a natural transformation $\Theta: F' \Rightarrow F$ such that



Since the 2-morphisms are given by natural transformations with an additional property, we have that quasi Q-groupoids form a strict 2-category as claimed at the beginning of the section.

3. Degree one quasi Q-groupoids: quasi LA-groupoids

In this section we focus on 1-graded quasi Q-groupoids. We show how these objects are a common generalization of LA-groupoids and L_2 -algebroids. Also, we prove that these objects are equivalent to linear quasi Poisson structures on the corresponding dual VB-groupoids and give interesting examples that motivate their study. Based on Corollary 2.16, we show that these objects provide us with the appropriate Morita invariant concept of Lie algebroid. Therefore we can define a Lie algebroid in the category of differentiable stacks as the Morita class of a 1-graded quasi Q-groupoid.



3.1. The functor [1]. In this section we summarize some well known facts about vector bundles and 1-manifolds that we will use later. Some standard references are [70, 72, 85, 90].

Denote by \mathcal{VB} the category of vector bundles and by 1- $\mathcal{M}an$ the category of degree 1manifolds. In Example 2.1 we show that, given a vector bundle $E \to M$, one can produce the 1-manifold $E[1] = (M, \Gamma \wedge^{\bullet} E^*)$. It is not difficult to see that this construction is functorial. Indeed, the functor

$$[1]: \mathcal{VB} \to 1\text{-}\mathcal{Man}, \quad E \to M \rightsquigarrow E[1]$$

is an equivalence of categories. Therefore, the study of 1-manifolds is equivalent to the study of vector bundles. In what follows we recall how this functor can be upgraded to include Lie algebroids [90] and LA-groupoids [70, 72].

3.1.1. Vector fields on 1-manifolds. Recall that a Lie algebroid is a vector bundle $A \to M$ with a vector bundle map $\rho : A \to TM$, called the anchor, and a bracket $[\cdot, \cdot] : \Gamma A \times \Gamma A \to \Gamma A$ satisfying

(13)
$$[a,b] = -[b,a], \quad [a,fb] = f[a,b] + \rho(a)(f)b,$$

(14)
$$[a, [b, c]] = [[a, b], c] + [b, [a, c]],$$

for all $a, b, c \in \Gamma A$ and $f \in C^{\infty}(M)$. If only the equations in (13) are satisfied then we say that $(A \to M, [\cdot, \cdot], \rho)$ is a *pre-Lie algebroid*, see [43, Def. 2.1] or [42, Def. 3.1] where pre-Lie algebroids are called *skew algebroids*.

A pre-Lie algebroid is the same as a linear $\pi \in \mathfrak{X}^2(A^*)$. On the other hand, Equation (14) is equivalent to $[\pi,\pi] = 0$ [37], see e.g. [42, Thm. 3.1]; the same result also shows that $Q \in \mathfrak{X}^{1,1}(A[1])$ determines a pre-Lie algebroid structure on $A \to M$ and equation (14) is satisfied if and only if $Q^2 = 0$, see also [90]. The relation between $([\cdot, \cdot], \rho)$ and Q is given by the usual Chevalley-Eilenberg formula for $\omega \in C^k(A[1])$

(15)
$$Q(\omega)(a_0, \cdots a_k) = \sum_{i=0}^k (-1)^i \rho(a_i)(\omega(a_0, \cdots, \widehat{a}_i, \cdots)) + \sum_{i < j} (-1)^{i+j} \omega([a_i, a_j], a_0, \cdots, \widehat{a}_i, \cdots, \widehat{a}_j, \cdots, a_k)$$

Thus the category of Lie algebroids and the category of degree 1 Q-manifolds are equivalent [90]. One can extend the above observations to multivector fields [50, Prop. 3.7], see also [22, §6.1].

Proposition 3.1. Let $A \to M$ be a vector bundle and take a multivector field on A^* , $\beta \in \mathfrak{X}^n(A^*)$. Then the following are equivalent:

- (1) β is a linear multivector field,
- (2) the contraction $\beta : \bigoplus_{A^*}^n T^*A^* \to \mathbb{R}$ is a vector bundle map over $\bigoplus_{A^*}^n A$,
- (3) β is equivalent to $\sigma_{\beta} \in \mathfrak{X}^{1,n-1}(A[1])$, which is defined by $\sigma_{\beta}(\omega)^{\uparrow} = [\beta, \omega^{\uparrow}]$ for $\omega \in \Gamma \wedge^{\bullet} A^{*}$, where $(\cdot)^{\uparrow}$ denotes the vertical lift as in [67, Eq.(3)].

Moreover, the assignment $\beta \to \sigma_{\beta}$ satisfies $\sigma_{[\beta_1,\beta_2]} = [\sigma_{\beta_1},\sigma_{\beta_2}]$.

3.1.2. LA-groupoids as Q-groupoids. Here we collect some basic facts about groupoids in the category \mathcal{VB} . Recall that a VB-groupoid $(H \rightrightarrows E; G \rightrightarrows M)$ is a diagram of groupoids and vector bundles like (16) such that $\operatorname{Gr}(\mathfrak{m}_H) \to \operatorname{Gr}(\mathfrak{m}_G)$ is a vector subbundle of $H \times H \times H \to G \times G \times G$, see e.g. [44, 66] for equivalent definitions and other properties.

The core of a VB-groupoid is the vector bundle over M defined by $C = \mathfrak{u}_G^* \ker(\mathfrak{s}_H) \to M$. One can show that the vector bundle $H^* \to G$ inherits a VB-groupoid structure $(H^* \rightrightarrows C^*; G \rightrightarrows M)$ with core $E^* \to M$ known as the *dual VB-groupoid* [66, §11.2]. As Lie groupoids differentiate to Lie algebroids, the infinitesimal counterpart of a VB-groupoid is a *VB-algebroid*. We denote the VB-algebroid of (16) by (17), we follow the convention from [23] according to which double arrows indicate a Lie algebroid structure over a manifold. VB-groupoids and VB-algebroids form a category with morphisms defined in the natural way.

$$(17) \qquad A_H \Longrightarrow E \\ \downarrow \qquad \downarrow \\ A_C \Longrightarrow M$$

In this work we will need an additional structure on the vertical bundles in the diagram (16).

Definition 3.2. $(H \Rightarrow E; G \Rightarrow M)$ is an *LA-groupoid* if $(H \Rightarrow E; G \Rightarrow M)$ is a VB-groupoid and $(H \rightarrow G, [\cdot, \cdot], \rho)$ is a Lie algebroid such that $\operatorname{Gr}(\mathfrak{m}_H) \rightarrow \operatorname{Gr}(\mathfrak{m}_G)$ is a Lie subalgebroid of $H \times H \times H \rightarrow G \times G \times G$, see [59, Def. 2.1].

If $(H \to G, [\cdot, \cdot], \rho)$ is just a pre-Lie algebroid, see §3.1.1, then we refer $(H \rightrightarrows E; G \rightrightarrows M)$ as a pre-LA-groupoid.

LA-groupoids have a rich Lie theory, see e.g. [23]. Their double nature allows us to ask questions about integration and differentiation. Let us mention here that their infinitesimal counterparts are double Lie algebroids whereas their global counterparts are double Lie groupoids $\S A$.

Another property that we want to emphasize is the duality between LA-groupoids and PVBgroupoids displayed in diagram (18). In other words, $(H \Rightarrow E; G \Rightarrow M)$ is an LA-groupoid if and only if $(H^* \Rightarrow C^*; G \Rightarrow M)$ is a *PVB-groupoid*, i.e. H^* has a linear and multiplicative Poisson structure.

(18) LA-groupoid
$$\begin{array}{c} H \Longrightarrow E \\ \downarrow \\ G \Longrightarrow M \end{array} \xrightarrow{dual} \begin{array}{c} H^* \Longrightarrow C^* \\ \downarrow \\ G \Longrightarrow M \end{array}$$
 PVB-groupoid.

Using the functor [1] one can translate the above definitions into supergeometric concepts. This was done in [70, 71, 72], we summarize here the relevant results for us in the following proposition.

Proposition 3.3 (See [70, 71, 72]). The functor [1] has the following the properties:

- (1) It sends the VB-groupoid $(H \rightrightarrows E; G \rightrightarrows M)$ to the 1-graded Lie groupoid $H[1] \rightrightarrows E[1]$.
- (2) Pre-LA-groupoid structures on $(H \rightrightarrows E; G \rightrightarrows M)$ are in one to one correspondence with $Q \in \mathfrak{X}^{1,1}_{mul}(H[1]).$
- (3) The diagram

$$\{LA\text{-}groupoids\} \xrightarrow{[1]} \{1\text{-}graded Q\text{-}groupoids\}$$

$$\downarrow_{Lie} \qquad \qquad \downarrow_{Lie} \qquad \qquad \downarrow_{Lie}$$

$$\{Double \ Lie \ algebroids\} \xrightarrow{[1]} \{1\text{-}graded \ Q\text{-}algebroids\}$$

is commutative and the horizontal arrows are equivalences of categories.

3.2. Quasi LA-groupoids. Motivated by Proposition 3.3, we can make the following definition.

Definition 3.4. Let $(H \rightrightarrows E, G \rightrightarrows M)$ be a VB-groupoid. We say that $(H \rightrightarrows E, G \rightrightarrows M)$ is a quasi LA-groupoid if $H[1] \rightrightarrows E[1]$ is a quasi Q-groupoid.

Since the quasi Q-groupoids introduced here are a generalization of Q-groupoids. We obtain that quasi LA-groupoids, i.e. 1-graded quasi Q-groupoids, are a generalization of LA-groupoids. Some immediate consequences of the definition are the following.

- (1) Quasi LA-groupoids form a strict 2-category with morphisms given as in §2.5.
- (2) Given VB-groupoids $(H^i \Rightarrow E^i; G^i \Rightarrow M^i)$ $i \in \{1, 2\}$, they are VB-Morita equivalent as defined in [39, §3] if and only if $H^1[1] \Rightarrow E^1[1]$ and $H^2[1] \Rightarrow E^2[1]$ are Morita equivalent as 1-graded Lie groupoids. Thus, Corollary 2.16 implies that quasi LA-groupoid structures can be transported along VB-Morita equivalences.
- (3) Therefore, we can make the following definition. A Lie algebroid in the category of differentiable stacks is the Morita class of a quasi LA-groupoid. Compare with [94].

Let $(H \rightrightarrows E, G \rightrightarrows M)$ be a VB-groupoid with VB-algebroid $(A_H \Rightarrow E; A_G \Rightarrow M)$. Recall that a quasi Q-groupoid structure on $H[1] \rightrightarrows E[1]$ is given by a vector field $Q \in \mathfrak{X}^{1,1}_{mul}(H[1])$ and a section $q \in \Gamma^2(\mathcal{A}_{H[1]}) = \Gamma^2(A_H[1] \rightarrow E[1])$ satisfying

(19)
$$Q^2 = q^l - q^r$$
 and $[Q, q^r] = 0.$

By Proposition 3.3, we know that Q is equivalent to a pre-LA-groupoid structure on $(H \Rightarrow E, G \Rightarrow M)$, i.e. a bracket $[\cdot, \cdot]$ and an anchor ρ in H satisfying that the graph of the multiplication is a subalgebroid. We still have to interpret $q \in \Gamma^2(A_H[1])$ and (19).

Denote by $\Omega \to M$ the dual of the vector bundle whose space of sections are the double linear functions on A_H . It is easy to see (for example using the geometrization functor given in [26]) that a section $q \in \Gamma^2(A_H[1])$ is equivalent to two vector bundle maps

$$\lambda : \wedge^2 E \to A_G \text{ and } \Psi : \wedge^3 E \to \Omega$$

satisfying

$$\operatorname{pr}\circ\Psi(e_1\wedge e_2\wedge e_3) = \frac{1}{3}\Big(\lambda(e_1\wedge e_2)\otimes e_3 - \lambda(e_1\wedge e_3)\otimes e_2 + \lambda(e_2\wedge e_3)\otimes e_1\Big)$$

for $e_1, e_2, e_3 \in E$ and pr : $\Omega \to A_G \otimes E$. Equations (19) can be expressed in terms of $([\cdot, \cdot], \rho, \lambda, \Psi)$ but the resulting characterization is not particularly useful except in some special situations that we shall describe below. Nevertheless, we can say that λ controls the failure of the anchor map ρ to be bracket preserving whereas Ψ controls the failure of the Jacobi identity for the bracket.

3.2.1. *Quasi PVB-groupoids and duality.* The aim of this section is to generalize the duality of diagram (18) to quasi LA-groupoids.

Given a double vector bundle $(D \to B; A \to M)$, consider the vector bundle $\wedge^k D^* \to B$. We say that $\omega \in \Gamma(\wedge^k D^*)$ is *linear* if

$$\omega(s_1, \cdots, s_k) \in C_{lin}^{\infty}(B), \quad \omega(c_1, s_2, \cdots, s_k) \in C^{\infty}(M) \quad \text{and} \quad \omega(c_1, c_2, d_3, \cdots, d_k) = 0,$$

for $s_i \in \Gamma_{lin}(D, B)$ linear sections, $c_1, c_2 \in \Gamma_{cor}(D)$ core sections and $d_i \in \Gamma(D, B)$ arbitrary sections.

Definition 3.5. A quasi *PVB*-groupoid is a VB-groupoid $(H \Rightarrow E; G \Rightarrow M)$ endowed with $\Pi \in \mathfrak{X}^2_{mul}(H)$ and $\pi \in \Gamma \wedge^3 (A_H \to E)$ such that Π and π are linear sections and satisfy the equations (9).

Now we can state the main result of this section which generalizes [57, Thm. 5.9].

Theorem 3.6. Let $(H \rightrightarrows E; G \rightrightarrows M)$ be a VB-groupoid with core C. Then the correspondence of Proposition 3.1 determines the following duality

$$\begin{array}{cccc} quasi \ LA-groupoid & H \Longrightarrow E & H^* \Longrightarrow C^* \\ \parallel & \parallel & \parallel & dual & \downarrow & \downarrow & \\ \psi & \psi & & \downarrow & \downarrow & \downarrow & \\ G \Longrightarrow M & G \Longrightarrow M & G \Longrightarrow M \end{array} quasi \ PVB-groupoid;$$

which extends the one of (18).

We shall prove this theorem by relying on the description of infinitesimal derivations in terms of classical differential geometry proven in Proposition B.1. We give another proof of Theorem 3.6 using a different idea in Proposition 4.6.

Proof of Theorem 3.6. Let Q be a multiplicative vector field of degree 1 on the graded groupoid $H[1] \rightrightarrows E[1]$, so it is equivalent to a differential of degree 1 on $\Gamma(\wedge^{\bullet}H^*)$. According to Proposition 3.1, Q corresponds to a linear bivector field Π on H^* . The fact that Q is multiplicative is equivalent to the multiplicativity of Π with respect to the dual VB-groupoid $H^* \rightrightarrows C^*$. Note that the graded Jacobi identity implies that $Q^2(\cdot)^{\uparrow} = \frac{1}{2}[[\Pi,\Pi], \cdot^{\uparrow}]$. Since Proposition B.1 implies that any $q \in \Gamma^2 \mathcal{A}_{H[1]} = \Gamma^2(\mathcal{A}_H[1])$ satisfies $q^r(\cdot)^{\uparrow} = [\pi^r, \cdot^{\uparrow}]$ for some linear section π , we have that the condition $Q^2 = q^l - q^r$ is equivalent to $\frac{1}{2}[\Pi,\Pi] = \pi^l - \pi^r$. In such a situation, applying Proposition 3.1 to the right invariant derivation $[Q, q^r]$ allows us to conclude that $[Q, q^r] = 0$ is equivalent to $[\Pi, \pi^r] = 0$ (notice that the commutator of derivations corresponds to the Schouten bracket of linear multivector fields).

3.2.2. Lie theory of quasi LA-groupoids. The infinitesimal counterpart of a quasi LA-groupoid is a generalization of the double Lie algebroids introduced in [64]. Recall from [50] that a Lie quasibialgebroid is a triple $(A \Rightarrow M, \delta, \psi)$, where $A \Rightarrow M$ is a Lie algebroid, $\delta : \Gamma(\wedge^{\bullet} A) \to \Gamma(\wedge^{\bullet+1} A)$ is a degree 1 derivation and $\psi \in \Gamma(\wedge^{3} A)$ satisfying:

$$\delta([\cdot, \cdot]) = [\delta \cdot, \cdot] + [\cdot, \delta \cdot], \quad \delta^2(\cdot) = [\psi, \cdot] \text{ and } \delta(\psi) = 0.$$

Definition 3.7. A quasi LA-algebroid $((D \Rightarrow E; A \Rightarrow M), (\mathbf{a}^A, [,]^A), \pi)$ is given by a VBalgebroid $(D \Rightarrow E; A \Rightarrow M)$ with bracket and anchor $(\mathbf{a}^E, [,]^E)$, a linear pre-Lie algebroid structure $(\mathbf{a}^A, [\cdot, \cdot]^A)$ on $(D \to A; E \to M)$ and $\pi \in \Gamma(\wedge^3 D^{\uparrow} \to C^*)$ a linear section with the following properties:

- (1) the anchor \mathbf{a}^E defines a morphism of pre-Lie algebroids: $(D \to A; E \to M) \to (TE \to TM; E \to M)$, where $TE \to TM$ is the tangent prolongation of $E \to M$,
- (2) the anchor \mathbf{a}_A defines a morphism of Lie algebroids $(D \to E; A \to M) \to (TA \to TM; A \to M)$, where $TA \to TM$ is the tangent prolongation of $A \to M$,
- (3) there is a Lie quasi-bialgebroid structure $(D^{\uparrow}, \delta_D, \pi)$ on the VB-algebroid $(D^{\uparrow} \Rightarrow C^*; A \Rightarrow M)$ dual to $(D \Rightarrow E; A \Rightarrow M)$ see [66, §9.2], where the differential δ_D on $\wedge^{\bullet} D^{\uparrow} \Rightarrow C^*$ is given by $\delta_D(\cdot)^{\uparrow} = [\Pi_A, \cdot^{\uparrow}]$ and $\Pi_A \in \mathfrak{X}^2(D^{\uparrow})$ is the bivector field associated to $(\mathbf{a}^A, [,]^A)$.

Remark 3.8. The previous definition is inspired on [64]. Alternatively, one can introduce a quasi LA-algebroid as a VB-algebroid such that applying the functor [1] to the vertical bundle structures produces a quasi Q-algebroid. The equivalence between these two definitions is a long exercise, see [70, §4.2.2]. See [91] for a similar situation.

It is proven in [65, §1] that an LA-groupoid differentiates to a double Lie algebroid. Similarly, we can differentiate a quasi LA-groupoid and obtain a quasi LA-algebroid. We can prove this result following the approach of [65]. Alternatively, we can dualize the involved structures and apply Theorem 3.6 which is the approach we prefer.

Definition 3.9. A quasi *PVB-algebroid* is a VB-algebroid $(D \Rightarrow E; A \Rightarrow M)$ endowed with a linear $\Pi \in \mathfrak{X}^2(D)$ and a linear $\pi \in \Gamma(\wedge^3 D \to E)$ such that $(D \Rightarrow E, \sigma_{\Pi}, \pi)$ (as in Proposition 3.1) is a Lie quasi bialgebroid.

Remark 3.10. Observe that in the previous definition $TD \to D$ is a double vector bundle in two different ways $(TD \to D; TE \to E)$ and $(TD \to D; TA \to A)$ and we demand that $\pi \in \mathfrak{X}^2(D)$ is linear with respect to both structures.

Some immediate consequences of the definitions given above are the following.

- (1) The vertical dual of a quasi LA-algebroid is a quasi PVB-algebroid.
- (2) Let $((H \Rightarrow E, G \Rightarrow M), \Pi, \pi)$ be a quasi PVB-groupoid. Then $((A_H \Rightarrow E, A_G \Rightarrow M), Lie(\Pi), \pi)$ is a quasi PVB-algebroid.
- (3) As observed before, this implies the differentiation of quasi LA-groupoids.

Theorem 3.11. The Lie functor establishes a bijection between the set of compatible quasi LAgroupoid structures on a source-simply connected VB-groupoid and the set of quasi LA-algebroid structures compatible with the corresponding VB-algebroid. \Box

Using the dualization provided by Theorem 3.6, this result can be proven following the ideas in [23, Prop. 5.3.1] and [23, Thm. 5.3.5]. In fact, the same arguments that hold for linear Poisson groupoids structures carry over verbatim to linear quasi-Poisson groupoid structures.

Remark 3.12 (Lie groupoids in the category of stacks). Recall that an LA-groupoid can be regarded as the Lie algebroid of a double Lie groupoid, see e.g. [63]. Hence, quasi LA-groupoids should have a global counterpart. If we think of quasi LA-groupoids as Lie algebroids in the category of differentiable stacks, we expect that they should integrate to Lie groupoids in the category of differentiable stacks which are objects that generalize double Lie groupoids, see [97, 98] for some work in this direction.

3.3. L_2 -algebroids. In this section we extend the main result of [9] by showing that the strict 2-category of quasi LA-groupoids over a unit groupoid is equivalent to the strict 2-category of L_2 -algebroids.

Recall that a L_2 -algebroid, see e.g. [15], consist of two vector bundles over the same base $E \to M$ and $C \to M$ endowed with

(20)
$$\rho: E \to TM$$
, $\partial: C \to E$, $[\cdot, \cdot]: \Gamma E \wedge E \to \Gamma E$, $\nabla: \Gamma E \otimes C \to \Gamma C$, $K: \wedge^3 E \to C$
satisfying the Leibniz rules

satisfying the Leibniz rules

(21)
$$[e, fe'] = f[e, e'] + \rho(e)(f)e', \quad \nabla_{fe}c = f\nabla_e c, \quad \nabla_e(fc) = f\nabla_e c + \rho(e)(f)c$$

for $e, e' \in \Gamma E$, $c \in \Gamma C$ and $f \in C^{\infty}(M)$ and the integrability conditions

(22)
$$\rho(\partial c) = 0, \quad \partial \nabla_e c = [e, \partial c], \quad \nabla_{\partial c_1} c_2 = -\nabla_{\partial c_2} c_1$$

(23)
$$\partial K(e_1, e_2, e_3) = [e_1, [e_2, e_3]] - [[e_1, e_2], e_3] + [[e_1, e_3], e_2]$$

(24)
$$K(\partial c, e_1, e_2) = [\nabla_{e_1}, \nabla_{e_2}]c - \nabla_{[e_1, e_2]}c,$$

(25)
$$\left(\nabla_{e_1} K(e_2, e_3, e_4) + (-1)^i c. p. \right) = \left(K([e_1, e_2], e_3, e_4) + (-1)^i c. p. \right) + K([e_2, e_4], e_1, e_3) + K([e_1, e_3], e_2, e_4),$$

for $e_i \in \Gamma E$ and $c_i \in \Gamma C$.

By relying on the supergeometric viewpoint, we get a straightforward conceptual description of the 1-morphisms between L_2 -algebroids.

Proposition 3.13 (See [15]). The functor

$$[1]: \{L_2\text{-}algebroids\} \rightarrow \{Degree \ 2 \ Q\text{-}manifolds\}$$

is an equivalence of categories

Remark 3.14. It is proven in [60] that degree 2 Q-manifolds are also equivalent to VB-Courant algebroids. Therefore, all the results obtained in this sections can ve re-interpret in terms of them.

Thus, in the supergeometric setting we get that a 1-morphism is a morphism of degree 2 Q-manifolds while a 2-morphism between two degree 2 Q-manifold morphisms is given by a Q-homotopy. Using the equivalence described above, we can express morphisms between L_2 -algebroids in a straightforward but laborious process. For the sake of clarity, we restrict here to base preserving morphisms, for the general case we refer to [15].

A morphism between L_2 -algebroids $(\partial : C \to E, \rho, \nabla, [\cdot, \cdot], K)$ and $(\partial' : C' \to E', \rho', \nabla', [\cdot, \cdot]', K')$ over the same base manifold M is a triple of vector bundle morphisms $F_0 : E \to E', F_1 : C \to C'$ and $\beta : \wedge^2 E \to C'$ such that

(26)
$$F_0 \circ \partial = \partial' \circ F_1,$$

(27)
$$\rho' \circ F_0 = \rho, \quad F_0([e_1, e_2]) = [F_0(e_1), F_0(e_2)]' + \partial'(\beta(e_1, e_2))$$

(28)
$$F_1(\nabla_e c) - \beta(e, \partial(c)) = \nabla'_{F_0(e)} F_1(c)$$

(29)
$$F_1(K(e_1, e_2, e_3)) + (\beta([e_1, e_2], e_3) + c.p.) = K'(F_0(e_1), F_0(e_2), F_0(e_3)) + (\nabla'_{F_0(e_1)}\beta(e_2, e_3) + c.p.);$$

for all $e, e_i \in \Gamma E$ and all $c \in \Gamma C$. The composition of L_2 -algebroid morphisms is defined in the natural way. Finally, a 2-morphism θ between the two L_2 -algebroid morphisms

$$(F_0, F_1, \beta), (G_0, G_1, \gamma) : (\partial : C \to E, \rho, \nabla, [\cdot, \cdot], K) \to (\partial' : C' \to E', \rho', \nabla', [\cdot, \cdot]', K')$$

is a vector bundle morphism $\theta: E \to C'$ which is a chain homotopy between the chain complex morphisms (F_0, F_1) and (G_0, G_1) and is such that

(30)
$$\beta(e_1, e_2) - \gamma(e_1, e_2) = \nabla'_{F_0(e_1)} \theta(e_2) - \nabla'_{F_0(e_2)} \theta(e_1) - \theta([e_1, e_2]) \quad \forall e_1, e_2 \in \Gamma E.$$

Thus L_2 -algebroids form a strict 2-category that we denote by L_2 -Alg. In particular, if M is a point, we recover the strict 2-category of L_2 -algebras (2-term L_{∞} -algebras) defined in [9, §4.3].

As we have pointed out, quasi LA-groupoids form a strict 2-category that we denote by \mathbf{qLA} . And let us denote by \mathbf{qLA}_u the full sub 2-category whose objects are just the quasi LA-groupoids that are VB-groupoids over a unit groupoid.

Theorem 3.15. The strict 2-categories L_2 -Alg and \mathbf{qLA}_u are 2-equivalent.

Proof. Consider a L_2 -algebroid given by (20). Define the VB-groupoid $(C \oplus E \rightrightarrows E, M \rightrightarrows M)$ with

(31)
$$\mathbf{s}_H(c,e) = e, \quad \mathbf{t}_H(c,e) = e + \partial c, \quad \mathbf{u}_H(e) = (0,e), \quad \mathbf{m}_H((c',e+\partial c),(c,e)) = (c+c',e).$$

Define a bracket and an anchor on $H = C \oplus E$ by means of

$$\rho_H(c,e) = \rho(e), \quad [(c,e), (c',e')]_H = (\nabla_e c' - \nabla_{e'} c + \nabla_{\partial c} c', [e,e'])$$

for $(c, e), (c', e') \in \Gamma H$. One can easily verify that $(H, [\cdot, \cdot]_H, \rho_H)$ is a pre-Lie algebroid due to (21) and that

$$\operatorname{Gr}(\mathbf{m}_H) \to \operatorname{Gr}(\mathbf{m}_M) \subset H \times H \times H \to M \times M \times M$$

is a subalgebroid due to (22). Thus, we get that $H[1] \Rightarrow E[1]$ has $Q \in \mathfrak{X}_{mul}^{1,1}(H[1])$ given by (15).

Since the VB-algebroid of $(H \rightrightarrows E, M \rightrightarrows M)$ is $(C \oplus E \Rightarrow E; 0_M \Rightarrow M)$, it is straightforward to verify that $K : \wedge^3 E \to C$ determines exactly a $q \in \Gamma^2(\mathcal{A}_{H[1]}) = \Gamma^2(A_H[1])$. The groupoid structure on H allows us to describe q^r and q^l explicitly as well. Given $f \in C^{\infty}(M)$, $\theta \in C^1(H[1]) = \Gamma H^*$ and $h_i = (c_i, e_i) \in \Gamma H$ for i = 0, 1, 2 we get

(32)

$$q^{r}(f) = a^{l}(f) = 0,$$

$$q^{r}(\theta)(h_{0}, h_{1}, h_{2}) = \langle \theta, \left(K(e_{0} + \partial c_{0}, e_{1} + \partial c_{1}, e_{2} + \partial c_{2}), 0 \right) \rangle,$$

$$q^{l}(\theta)(h_{0}, h_{1}, h_{2}) = \langle \theta, \left(K(e_{0}, e_{1}, e_{2}), -\partial K(e_{0}, e_{1}, e_{2}) \right) \rangle.$$

Hence, a straightforward computation shows that equations (23) and (24) imply that $Q^2 = q^l - q^r$ and equation (24) implies $[Q, q^r] = 0$.

Conversely, let $(H \rightrightarrows E, M \rightrightarrows M)$ a VB-groupoid with core C such that $(H[1] \rightrightarrows E[1], Q, q)$ is a 1-graded quasi Q-groupoid. The unit subbundle induces a splitting of the exact sequence

$$0 \to C \to H \xrightarrow{\mathbf{s}_H} E \to 0,$$

thus $H = C \oplus E$ and the target $\partial := t_{|C} : C \to E$ is a vector bundle map. In terms of this splitting, the structure maps become the ones given in (31).

By Proposition 3.3 we have that $Q \in \mathfrak{X}_{mul}^{1,1}(H[1])$ is equivalent to a pre-Lie algebroid structure $([\cdot, \cdot]_H, \rho_H)$ on H satisfying

(33)
$$\operatorname{Gr}(\mathfrak{m}_H) \to \operatorname{Gr}(\mathfrak{m}_M) \subset H \times H \times H \to M \times M \times M$$

is a subalgebroid. Thus we can define

$$[e, e'] = [(0, e), (0, e')]_H, \quad \nabla_e c = [(0, e), (c, 0)]_H, \quad \rho(e) = \rho_H(0, e).$$

This definition satisfies (21) because $\rho_H(0,c) = 0$. Indeed, this follows from the fact that $(TM \Rightarrow TM, M \Rightarrow M)$ has zero core. Moreover (33) implies that equations (22) hold. As before, the VB-algebroid of $(H \Rightarrow E, M \Rightarrow M)$ is $(C \oplus E \Rightarrow E; 0_M \Rightarrow M)$. Thus $q \in \Gamma^2(A_H[1])$ is given exactly by a bundle map $K : \wedge^3 E \to C$. Since the groupoid structure is the same as before, (32) still holds and then we have that $Q^2 = q^l - q^r$ is equivalent to (23) and (24), whereas $[Q, q^r] = 0$ is equivalent to (25).

Let us consider the equivalence at the level of 1-morphisms. Consider L_2 -algebroids ($\partial : C \to E, \rho, \nabla, [\cdot, \cdot], K$) and ($\partial' : C' \to E', \rho', \nabla', [\cdot, \cdot]', K'$) over M with respective 1-graded quasi Q-groupoids $(H[1] \rightrightarrows E[1], Q, q)$ and $(H'[1] \rightrightarrows E[1], Q', q')$ as before. In order to simplify the computations, we consider morphisms of type $(F_0, F_1, 0)$ and $(\mathrm{Id}, \mathrm{Id}, \beta)$, the general situation reduces to these two cases, thanks to Lemma 3.16.

For the morphism $(F_0, F_1, 0)$, equation (26) implies that

$$F_0: E[1] \to E'[1]$$
 with $F_1 \oplus F_0: H[1] \to H'[1]$

defines a graded groupoid morphism. Since $\beta = 0$, equations (27) and (28) implies that $F_1 \oplus F_0$ is a *Q*-morphism, thus Id : $T[1](F_1 \oplus F_0) \circ Q_{H[1]} \Rightarrow Q_{H'[1]} \circ (F_1 \oplus F_0)$ is a natural transformation and it is clear that (29) becomes then (11).

For the morphism (Id, Id, β), we need H[1] = H'[1]. By the previous discussion, we get a morphism of the form (Id, τ). Using Proposition 2.22, we need to check that β defines a $b \in \Gamma^1(A_H[1])$ which relates the quasi *Q*-structures as a gauge transformation. Since $A_H[1]$ is $(C \oplus E \Rightarrow E; 0_M \Rightarrow M)$, one can show that $\beta : \wedge^2 E \to C$ gives exactly such a $b \in \Gamma^1(A_H[1])$. Notice that $A_H[1]$ has zero bracket and b^r and b^l are given by

$$b^{r}(f) = b^{l}(f) = 0, \quad b^{r}(\theta)(h_{0}, h_{1}) = \langle \theta, (\beta(e_{0} + \partial c_{0}, e_{1} + \partial c_{1}), 0) \rangle$$

$$b^{l}(\theta)(h_{0}, h_{1}) = \langle \theta, (\beta(e_{0}, e_{1}), -\partial K(e_{0}, e_{1})) \rangle.$$

for $f \in C^{\infty}(M)$, $\theta \in C^{1}_{H[1]} = \Gamma H^{*}$ and $h_{i} = (c_{i}, e_{i}) \in \Gamma H$ with i = 0, 1. Thus equations (27) and (28) are equivalent to $Q' = Q + b^{l} - b^{r}$, whereas (29) is equivalent to $q' = q + [Q, b^{r}]$.

Conversely, given 1-morphisms of the type (F, Id) and (Id, τ) we can produce the corresponding morphisms $(F_0, F_1, 0)$ and (Id, Id, β) just as before. The equivalence between 2-morphisms is left to the interested reader.

Lemma 3.16. Every morphism in **qLA** can be factored as a composite of a gauge transformation and a morphism of the form (*, 0).

Proof. Let (F, τ) be a quasi LA-groupoid morphism between the quasi Q-groupoids $(H[1] \Rightarrow E[1], Q, q)$ and $(H'[1] \Rightarrow E[1], Q', q')$ with corresponding VB groupoids $H \Rightarrow E, H' \Rightarrow E'$ with respective cores C and C'. Consider the gauge transformation $(1, -b) : (H[1] \times H'[1], e^b(Q \times Q', q \times q')) \to (H[1] \times H'[1], Q \times Q', q \times q')$ defined by any extension b of τ :

$$b: \wedge^2(E \times E') \to C \times C' \text{ with } b((e_1, F_0(e_1)) \land (e_2, F_0(e_2)) = -\tau(e_1 \land e_2) \text{ for all } e_1 \land e_2 \in \wedge^2 E.$$

Then we have that $(F, \tau) = (\operatorname{pr}_{H'[1]}, 0) \circ (1, -b) \circ (i_F, 0)$, where $i_F : H \to H \times H'$ is the inclusion of H as the graph of F.

Let us point out the following consequence of Proposition 3.13 and Theorem 3.15.

Corollary 3.17. The category of degree 2 Q-manifolds and the category of quasi LA-groupoids over a unit manifold are equivalent. \Box

3.4. The category of multiplicative sections. Let $(H \Rightarrow E, G \Rightarrow M)$ be a VB-groupoid with core C. As pointed out in [80, §3], there is a 2-term complex of vector spaces

(34)
$$\Gamma(C) \xrightarrow{o_H} \Gamma_{mul}(H), \quad u \mapsto u^r - u^l \quad \forall u \in \Gamma(C).$$

If $H \Rightarrow E$ is an LA-groupoid, it was shown in [80, §7.1] that the Lie bracket on sections naturally induces a crossed module structure on the above 2-term complex which then induces a strict L_2 -algebra as in § A. Here we show that if $H \Rightarrow E$ is equipped with a quasi LA-groupoid structure, (34) becomes an L_2 -algebra. We will give a conceptually simple proof of this result by combining Theorem 3.6 with some observations from [34]. In the subsequent discussion, we will use the notation $[\cdot, \cdot]_2$ and $[\cdot, \cdot, \cdot]_3$ to represent the 2-bracket and 3-bracket of a Lie 2-algebra, respectively.

Lemma 3.18 (See [34]). Let $(G \rightrightarrows M, \Pi, \pi)$ be a quasi Poisson groupoid and consider the 2-term complex

(35)
$$C^{\infty}(M) \xrightarrow{\delta} C^{\infty}_{mul}(G), \quad k \mapsto \mathbf{s}^* k - \mathbf{t}^* k \quad \forall k \in C^{\infty}(M).$$

Then the 2-bracket $[\cdot, \cdot]_2$ defined by

 $[f,g]_2 = \{f,g\} := \mathcal{L}_{\Pi^{\sharp}(df)}g, \quad \mathbf{s}^*[f,k]_2 = \{f,\mathbf{s}^*k\} \quad \forall f,g \in C^{\infty}_{mul}(G), \ k \in C^{\infty}(M)$

together with the 3-bracket

 $[f,g,h]_3 = \Theta(f,g,h), \quad where \quad i_{df \wedge dg \wedge dh} \pi^r = \texttt{t}^* \Theta(f,g,h), \quad \forall f,g,h \in C^\infty_{mul}(G);$

turn (35) into an L_2 -algebra.

Proof. One can prove this directly. Alternatively, one can include $C^{\infty}(M) \xrightarrow{\delta} C^{\infty}_{mul}(G)$ into $\Omega^{1}(M) \xrightarrow{\delta} \Omega^{1}_{mul}(G)$ using the de Rham differential and then use [34, Thm 3.1].

Proposition 3.19. Let $H \rightrightarrows E$ be a quasi LA-groupoid with core C. Then its corresponding 2-term complex (34) inherits a canonical L_2 -algebra structure.

Proof. By Theorem 3.6, the quasi LA-groupoid structure on $H \rightrightarrows E$ is equivalent to a linear quasi-Poisson structure on the dual VB-groupoid $H^* \rightrightarrows C^*$. Now let us observe that we can embed the 2-term complex (34) into the complex of multiplicative functions of $H^* \rightrightarrows C^*$ by taking each section to its corresponding linear function:



as in Lemma 3.18. By linearity of the quasi-Poisson structure on $H^* \rightrightarrows C^*$, the image of (34) is an L_2 -subalgebra.

Remark 3.20. Naturally, the 2-bracket $[\cdot, \cdot]_2$ on $\Gamma(C) \xrightarrow{\delta_H} \Gamma_{mul}(H)$ is determined by the pre-Lie algebroid structure given by Q and the 3-bracket $[\cdot, \cdot, \cdot]_3$ is given by the homotopy q but the argument above simplifies the proof that they determine an L_2 -algebra structure.

Recall that [80, Thm. 7.3] shows how Morita equivalences of LA-groupoids induce quasiisomorphism between the corresponding 2-term complexes as in (34). Here we extend that result to Morita equivalent quasi LA-groupoids by showing that their associated L_2 -algebra structures as in Proposition 3.19 are quasi-isomorphic.

Proposition 3.21. Let $(F, \tau) : (H[1], Q, q) \to (H[1], Q, \tilde{q})$ be a quasi LA-groupoid morphism such that F is a VB-Morita equivalence over $\phi : G \to \tilde{G}$. Then it induces a quasi-isomorphism between their corresponding L_2 -algebras.

Proof. A VB-Morita equivalence does not induce, in general, a morphism between the corresponding 2-term complexes. Instead, it determines a pair of quasi-isomorphisms as follows. We can transport the quasi LA-groupoid structure $(\widetilde{H}[1], \widetilde{Q}, \widetilde{q})$ to a quasi-LA groupoid structure on the pullback VB-groupoid structure over G to produce the following diagram

$$\left(\Gamma(C) \xrightarrow{\delta_H} \Gamma_{mul}(H)\right) \xrightarrow{F'} \left(\Gamma(\phi_0^* \widetilde{C}) \xrightarrow{\phi^* \delta_{\widetilde{H}}} \Gamma_{mul}(\phi^* \widetilde{H})\right) \xleftarrow{i} \left(\Gamma(\widetilde{C}) \xrightarrow{\delta_{\widetilde{H}}} \Gamma_{mul}(\widetilde{H})\right)$$

where F' and i are the naturally induced maps and these are quasi-isomorphisms according to [80, Prop. 5.4].

Let us verify now the compatibility of the quasi-isomorphism described above with the brackets. We can construct $(\phi^* \widetilde{H} \Rightarrow \phi_0^* \widetilde{E}, Q', a')$ in such a way that (F, τ) factors as the composite of $(p, \widetilde{\tau}) : (\phi^* \widetilde{H} \Rightarrow \phi^* \widetilde{H}, Q', a') \to (\widetilde{H} \Rightarrow \widetilde{H}, \widetilde{Q}, \widetilde{a})$ and (F', τ') , where $F' : H \to \phi^* \widetilde{H}$ is the map induced by F and p is the projection.

Let us argue that F' induces an L_2 -algebra morphism. In this case, we get a morphism of complexes



It follows from the natural transformation condition that τ' is determined by a vector bundle map $\tau'_0 : \wedge^2 E \to \phi_0^* \widetilde{C}$ such that

$$Q\left((F')^*(\xi)\right) - (F')^*(Q'\xi) = T^*\xi \quad \forall \xi \in \Gamma(\phi^* \widetilde{H}^*),$$

$$T : \oplus^2 H \to \phi^* \widetilde{H} \quad T(u,v) = \tau_0(\mathsf{t}_H(u), \mathsf{t}_H(v))^r - \tau_0(\mathsf{s}_H(u), \mathsf{s}_H(v))^l \quad \forall u, v \in H_g \quad g \in G.$$

As a consequence, the map $\Upsilon : \wedge^2 \Gamma_{mul}(H) \to \Gamma(\phi_0^* \widetilde{C})$ defined by $\Upsilon(X, Y) = -\tau_0(\mathsf{t}_H(X), \mathsf{t}_H(Y))$ for all $X, Y \in \Gamma_{mul}(H)$ satisfies that

$$\begin{aligned} F'([X,Y]_2) &- [F'(X),F'(Y)]_2 = \phi^* \delta_{\widetilde{H}}(\Upsilon(X,Y)) \\ F'([X,c]_2) &- [F'(X),F'(c)]_2 = \Upsilon(X,\phi^* \delta_H(c)) \quad \forall X,Y \in \Gamma_{mul}(H), \ c \in \Gamma(C). \end{aligned}$$

On the other hand, the coherence law (11) for (F', τ') implies equation

$$F'([X,Y,Z]_3) + ([\Upsilon(X,Y),Z]_2 + c.p.) = [F'(X),F'(Y),F'(Z)]_3 - (\Upsilon([X,Y]_2,Z) + c.p.))$$

for all $X, Y, Z \in \Gamma_{mul}(H)$, where 'c.p.' denotes cyclic permutations. As a consequence, F' together with Υ determine an L_2 -algebra morphism as in [9, Def. 34]. One can argue similarly for $(p, \tilde{\tau})$.

3.5. Lie bialgebroids over Lie groupoids. Here we study the Maurer-Cartan elements of the dgla introduced in §2.4 in the case n = 1; by doing so we obtain a general categorification of the concept of Lie bialgebroid which specializes to a number of constructions previously considered in the literature.

Let $(H \Rightarrow E, G \Rightarrow M)$ be a VB-groupoid and denote by $(A_H \Rightarrow E, A_G \Rightarrow M)$ its VBalgebroid. As explained in §3.1 the functor [1] gives rise to a 1-graded Lie groupoid $H[1] \Rightarrow E[1]$ with 1-graded Lie algebroid $A_H[1] \Rightarrow E[1]$. Consider the dgla $(\mathcal{V}^{\bullet}_{H[1]}, d, [\cdot, \cdot])$ constructed in §2.4, a degree 1 element in $\mathcal{V}^{\bullet}_{H[1]}$ is the same as a collection

$$\Psi = \left((f, F), (q, Q), t, (\Pi, \pi), (\Phi, \phi) \right) \in \mathcal{V}^1_{H[1]}$$

where $f \in C^4(E[1])$, $F \in C^3_{mul}(H[1])$, $q \in \Gamma^2(A_H[1])$, $Q \in \mathfrak{X}^{1,1}_{mul}(H[1])$, $t \in \Gamma^0(\wedge^2 A_H[1])$, $\Pi \in \mathfrak{X}^{2,-1}_{mul}(H[1])$, $\pi \in \Gamma^{-2}(\wedge^3 A_H[1])$, $\Phi \in \mathfrak{X}^{3,-3}_{mul}(H[1])$ and $\phi \in \Gamma^{-4}(\wedge^4 A_H[1])$. Moreover, Ψ is a Maurer-Cartan element, i.e. $d\Psi + \frac{1}{2}[\Psi, \Psi] = 0$, if it satisfies the following two sets of 5 + 6 equations

$f^{r} - f^{l} + [Q, F] = 0,$ $q^{r} - q^{l} + \frac{1}{2}[Q, Q] + [\Pi, F] = 0,$ $t^{r} - t^{l} + [\Pi, Q] + [\Phi, F] = 0,$ $\pi^{r} - \pi^{l} + \frac{1}{2}[\Pi, \Pi] + [Q, \Phi] = 0,$ $\phi^{r} - \phi^{l} + [\Pi, \Phi] = 0,$	$\begin{split} [F,q] + [Q,f] &= 0, \\ [F,t] + [Q,q] + [\Pi,f] &= 0, \\ [F,\pi] + [Q,t] + [\Pi,q] + [\Phi,f] &= 0, \\ [F,\phi] + [Q,\pi] + [\Pi,t] + [\Phi,q] &= 0, \\ [Q,\phi] + [\Pi,\pi] + [\Phi,t] &= 0, \\ [\Pi,\phi] + [\Phi,\pi] &= 0. \end{split}$
---	---

Let us explain a way to interpret these equations. Firstly, observe that each capital letter is accompanied by a lower letter, the lower letters determine correction terms for some equations involving the capital letters. The pairs (f, F) and (Φ, ϕ) give the "twisted up to homotopy" versions of the main "up to homotopy" structures (q, Q) and (Π, π) . The element t is somewhat mysterious to us, we can view it as a homotopy for the compatibility condition.

In the first table we summarize the structures that we codify when the base groupoid $G \rightrightarrows M$ is $* \rightrightarrows *$:

Ψ	Core	Name	Ref.
((0,0),(0,Q),0,(0,0),(0,0))	0	Lie algebra	
$((0,0), (0,Q), 0, (\Pi,0), (0,0))$	0	Lie bialgebra	
$((0,F),(0,Q),0,(\Pi,0),(0,0))$	0	quasi-Lie bialgebra	[51, 85]
$((0,F),(0,Q),0,(\Pi,0),(\Phi,0))$	0	proto-bialgebra	[51, 85]
((0,0),(0,Q),0,(0,0),(0,0))		strict L_2 -algebra	[9]
ig((0,0),(q,Q),0,(0,0),(0,0)ig)		L_2 -algebra	[9]
$((0,0),(0,Q),0,(\Pi,0),(0,0))$		strict Lie 2-bialgebra	[12, 36, 55]
$((0,0),(q,Q),0,(\Pi,\pi),(0,0))$		weak Lie 2-bialgebra	[36]
$((0,0), (q,Q), t, (\Pi,\pi), (0,0))$		2-term $L_{\infty}[0,1]$ -bialgebra	[12]
$((f, F), (q, Q), t, (\Pi, \pi), (\Phi, \phi))$		quasi weak Lie 2-bialgebra	[36, Rmk. 2.8]

The above table can be reproduced by replacing the base groupoid with $M \rightrightarrows M$; at the level of terminology, this amounts to adding *-oid* at the end of each name. Instead of doing that, we give directly the case of a general base:

Ψ	Name	Ref.
ig((0,0),(0,Q),0,(0,0),(0,0)ig)	LA-groupoid	
ig((0,0),(q,Q),0,(0,0),(0,0)ig)	quasi LA-groupoid	$\S{3.2}$
$((0,0),(0,0),0,(\Pi,0),(0,0))$	PVB-groupoid	
$((0,0),(0,0),0,(\Pi,\pi),(0,0))$	quasi PVB-groupoid	$\S{3.2.1}$
$ig((0,0),(0,Q),0,(\Pi,0),(0,0)ig)$	Lie-bialgebroid groupoid	[24]
$ig((0,0),(0,Q),0,(\Pi,\pi),(0,0)ig)$	quasi-Poisson LA-groupoid	§ A
$((0,F),(0,Q),0,(\Pi,0),(\Phi,0))$	multiplicative proto-bialgebroid	

We will analyse some of these objects at the global level in \S A.

3.6. **Examples.** As we have seen, quasi LA-groupoids include LA-groupoids and L_2 -algebroids as particular cases. Now we will some more concrete examples which illustrate the scope of our viewpoint.

3.6.1. Cotangent bundles of quasi Poisson groupoids. It was shown in [63] that the cotangent prolongation of a Poisson groupoid is an LA-groupoid. Here we extend that result by showing how the cotangent of a quasi Poisson groupoid inherits a quasi LA-groupoid structure.

Proposition 3.22. Let $(G \rightrightarrows M, \Pi, \pi)$ be a quasi Poisson groupoid with Lie algebroid A. Then $T^*[1]G \rightrightarrows A^*[1]$ with

$$\Pi_{can} \in \mathfrak{X}^{2,-1}_{mul}(T^*[1]G), \quad \Theta = \Pi \in C^2_{mul}(T^*[1]G) \quad and \quad \theta = \pi \in C^3(A^*[1])$$

is a degree 1 symplectic quasi Q-groupoid. Moreover, any degree 1 symplectic quasi Q-groupoid is of this type.

Proof. Clearly Π_{can} is a multiplicative and non-degenerate Poisson structure with bracket given by the Schouten bracket. Thus equations (9) become (10). For the moreover part recall that a degree 1 symplectic manifold is symplectomorphic to $(T^*[1]M, \Pi_{can})$. Therefore, any degree 1 symplectic groupoid is symplectomorphic to $T^*[1]G \Rightarrow A^*[1]$ and this concludes the proof. \Box

We can apply Theorem 3.6 to the quasi LA-groupoid determined by a quasi Poisson groupoid as in Proposition 3.22 to conclude the following result.

Corollary 3.23. The tangent prolongation of a quasi Poisson groupoid is canonically a quasi Poisson groupoid. \Box

As natural as this Corollary appears, we haven't found it in the literature. Another immediate consequence of Propositions 3.19 and 3.22 is the following.

Corollary 3.24 ([34]). Let $(G \rightrightarrows M, \Pi, \pi)$ be a quasi Poisson groupoid, then the 2-term complex

$$\delta: \Omega^1(M) \to \Omega^1_{mul}(G)$$

naturally inherits an L_2 -algebra structure. Moreover, a Morita equivalence of quasi Poisson groupoids induces a quasi-isomorphism in the corresponding 2-term complexes \Box

Remark 3.25. The explicit form of the 3-bracket can be deduced by following our previous arguments see [34, Thm. 3.1]. The L_2 -algebra structure above can be extended to the entire spaces of differential forms $\Omega^{\bullet}(M) \to \Omega^{\bullet}_{mul}(G)$ [34, Thm. 3.6]. Note that our use of Theorem 3.6 allows us to essentially reduce that construction to the case of (degree 0) functions.

3.6.2. Central extensions and prequantum algebroids. Recall that $(\mathbb{R} \Rightarrow 0, * \Rightarrow *)$ is a VBgroupoid with multiplication given by addition. The corresponding degree 1-groupoid is denoted by $\mathbb{R}[1] \rightrightarrows 0$. If $(H \rightrightarrows E, G \rightrightarrows M)$ is another VB-groupoid, their sum $(H \oplus \mathbb{R}_G \rightrightarrows E, G \rightrightarrows M)$ is again a VB-groupoid, in graded terms this is given by the cartesian product $H[1] \times \mathbb{R}[1] \Rightarrow E[1]$. The next result shows one way to construct quasi Q-groupoid structures on this product.

Proposition 3.26. Let $(H[1] \Rightarrow E[1], Q, q)$ be a degree 1 quasi Q-groupoid, $h \in C^2_{mul}(H[1])$ and $\xi \in C^{3}(E[1])$. If

(37)
$$Q(h) = \mathbf{s}_{H[1]}^* \xi - \mathbf{t}_{H[1]}^* \xi \quad and \quad Q(\mathbf{t}_{H[1]}^* \xi) = \mathbf{t}_{H[1]}^* Q_0(\xi) = q^r(h)$$

then $(H[1] \times \mathbb{R}[1] \rightrightarrows E[1], \widehat{Q}, \widehat{q})$ is a quasi Q-groupoid with

$$\widehat{Q}(h_0 + \lambda h_1) = Q(h_0) + h \wedge h_1 - \lambda Q(h_1), \widehat{q}^r(h_0 + \lambda h_1) = q^r(h_0) + (\mathfrak{t}^*_{H[1]}\xi) \wedge h_1 + \lambda q^r(h_1),$$

where $h_0, h_1 \in C^{\bullet}(H[1])$ and λ is the degree 1 coordinate function of $\mathbb{R}[1]$.

Proof. Observe that $\mathcal{A}_{H[1]\times\mathbb{R}[1]} = \mathcal{A}_{H[1]}\times\mathbb{R}[1]$. If λ denote the degree 1 coordinate function of $\mathbb{R}[1]$ then $\lambda^2 = 0$ and a degree k function in $H[1] \times \mathbb{R}[1]$ is given by $h_0 + \lambda h_1$ with $h_0 \in C^k(H[1])$ and $h_1 \in C^{k-1}(H[1])$. Thus one can easily check that $\widehat{Q} \in \mathfrak{X}^{1,1}(H[1] \times \mathbb{R}[1])$ and $\widehat{q} \in \Gamma^2 \mathcal{A}_{H[1] \times \mathbb{R}[1]}$. Notice that (\hat{Q}, \hat{q}) is a quasi Q-structure satisfying

$$\widehat{Q}_{|C^k(H[1])} = Q$$
 and $\widehat{q}_{|\mathcal{A}_{H[1]}} = q.$

Hence, to show the result we just need to check how (\hat{Q}, \hat{q}) acts on λ . The vector field \hat{Q} is multiplicative because

$$\mathbf{m}_{H[1]\times\mathbb{R}[1]}^*\widehat{Q}(\lambda) = \mathbf{m}_{H[1]}^*h = \mathrm{pr}_1^*h + \mathrm{pr}_2^*h = \mathrm{pr}_1^*\widehat{Q}(\lambda) + \mathrm{pr}_2^*\widehat{Q}(\lambda) = \widehat{Q}^{(2)}\mathbf{m}_{H[1]\times\mathbb{R}[1]}^*\lambda$$

due to the multiplicativity of h. The first equation in (37) implies that

$$\widehat{Q}^2(\lambda) = Q(h) = \mathbf{s}_{H[1]}^* \xi - \mathbf{t}_{H[1]}^* \xi = (\widehat{q}^l - \widehat{q}^r)(\lambda).$$

Finally,

$$[\widehat{Q}, \widehat{q}^r](\lambda) = \widehat{Q}(\widehat{q}^r(\lambda)) - \widehat{q}^r(\widehat{Q}(\lambda)) = Q(\mathsf{t}^*_{H[1]}\xi) - q^r(h) = 0$$

ation in (27)

by the second equation in (37).

Corollary 3.27 (Classification of central extensions). Let $(H[1] \Rightarrow E[1], Q, a)$ be a degree 1 quasi Q-groupoid and $(K[1] \rightrightarrows E[1], \widehat{Q}, \widehat{q})$ a central extension.

(1) The choice of an splitting σ for the exact sequence $0 \to \mathbb{R}[1] \to K[1] \to H[1] \to 0$ makes the central extension isomorphic to the one constructed in Proposition 3.26 with $h = \sigma^* \widehat{Q}(\lambda) \text{ and } \mathtt{t}^*_{H[1]} \xi = \sigma^* \widehat{q}^r(\lambda).$

(2) The central extensions defined by (h,ξ) and (h',ξ') are equivalent if and only if there is an element $b \in C^2(E[1])$ such that $\xi - \xi' = Q_0(b)$ and $h - h' = \mathbf{s}^*_{H[1]}b - \mathbf{t}^*_{H[1]}b$.

Proof. Observe that, since λ is a multiplicative function, then $h = \sigma^* \widehat{Q}(\lambda)$ and $\mathfrak{t}^*_{H[1]} \xi = \sigma^* \widehat{q}^r(\lambda)$ makes sense and so h is multiplicative. Let us verify the equations in (37)

$$\begin{aligned} Q(h) &= Q(\sigma^* Q(\lambda)) = \sigma^* Q^2(\lambda) = \sigma^* (\widehat{q}^l - \widehat{q}^r)(\lambda) = \mathbf{s}_{H[1]}^* \xi - \mathbf{t}_{H[1]}^* \xi, \\ Q(\mathbf{t}_{H[1]}^* \xi) &= Q(\sigma^* \widehat{q}^r(\lambda)) = \sigma^* \widehat{Q}(\widehat{q}^r(\lambda)) = \sigma^* \widehat{q}^r(\widehat{Q}(\lambda)) = q^r(\sigma^* \widehat{Q}(\lambda)) = q^r(h), \end{aligned}$$

where, in the second equality of each equation, we used that λ is central. The isomorphism part is left to the reader.

For the second item, note that by degree reasons a different choice of splitting gives the same h and ξ . Thus we can suppose that the automorphism of $H[1] \times \mathbb{R}[1]$ is of the form $(1, \tau)$. Therefore, Proposition 2.22 implies that we get a gauge transformation for some $b \in \Gamma^1 \mathcal{A}_{H[1] \times \mathbb{R}[1]}$. Since such section must preserve H[1], we get that $b : E[1] \to \mathbb{R}[2]$ thus $b \in C^2(E[1])$. Since $\mathbb{R}[1]$ is an abelian algebra we get that the gauge transformation equation becomes $\xi - \xi' = [Q_0, b] = Q_0(b)$ and $h - h' = \mathbf{s}^*_{H[1]}b - \mathbf{t}^*_{H[1]}b$.

Some relevant examples of the above construction are given by Atiyah algebroids of S^{1} -gerbes and equivariant S^{1} -gerbes. First we recall the case of an S^{1} -principal bundles $P \to M$. Here the Atiyah algebroid is the bundle $A_{P} = TP/S^{1}$ which fits into the exact sequence

$$0 \to \mathbb{R}_M \to A_P \to TM \to 0.$$

Any principal connection, i.e. a splitting of the above sequence, determines a Lie algebroid isomorphism $A_P \cong TM \oplus \mathbb{R}_M = A_\omega$ with bracket

$$[(X, f), (Y, g)] = ([X, Y], X(g) - X(f) - i_X i_Y \omega),$$

where $\omega \in \Omega^2_{cl}(M)$ is the curvature of the connection.

A relevant case of the above is given by the prequantum bundle of a symplectic manifold (M, ω) . Here one starts with (M, ω) , and thus with A_{ω} , and need to find an S^1 -principal bundle $P \to M$ and a principal connection making $A_P \cong A_{\omega}$. This problem was answered in [52], see also [38].

In what follows we give the analogues of A_{ω} for closed 3-forms on manifolds and closed 1shifted 2-forms on Lie groupoids (the latter case includes the former). The link with higher connections and Atiyah algebroids will be explained in § 3.6.5.

Example 3.28 (Closed 3-forms). Let M be a manifold, $H \in \Omega^3(M)$ with dH = 0 and consider the LA-groupoid $(TM \rightrightarrows TM, M \rightrightarrows M)$ then $(T[1]M \rightrightarrows T[1]M, Q = d_{dr})$ with h = 0 and $\xi = H$ satisfy the hypothesis of Proposition 3.26. Thus we get a quasi LA-groupoid on $(TM \oplus \mathbb{R}_M \rightrightarrows TM, M \rightrightarrows M)$. Using Theorem 3.15 one can show that this is exactly the L_2 -algebroid in $\mathbb{R}_M \rightarrow TM$ given by

$$\rho = \mathrm{Id}, \quad \partial = 0, \quad [X, Y] = [X, Y], \quad \nabla = 0 \quad \mathrm{and} \quad K(X, Y, Z) = i_X i_Y i_Z H$$

for $X, Y, Z \in \mathfrak{X}(M)$.

An extension of the previous example is given by *closed* 1-*shifted* 2-*forms* on Lie groupoids.

Example 3.29 (Closed +1 shifted 2-forms). Recall, see e.g. [25, 96], that a closed 1-shifted 2-form on a Lie groupoid $G \rightrightarrows M$ is an $\omega \in \Omega^2(G)$ and an $H \in \Omega^3(M)$ satisfying

$$\mathbf{m}^*\omega = \mathrm{pr}_1^*\omega + \mathrm{pr}_2^*\omega, \quad d\omega = \mathbf{s}^*H - \mathbf{t}^*H \quad \text{and} \quad dH = 0.$$

Then the quasi Q-groupoid $(T[1]G \Rightarrow T[1]M, Q = d_{dr})$ with $h = \omega$ and $\xi = H$ satisfy the hypothesis of Proposition 3.26. Thus we get the quasi LA-groupoid

$$(38) \begin{array}{c} TG \oplus \mathbb{R}_G \Longrightarrow TM \\ \parallel & \parallel \\ \Downarrow & \Downarrow \\ G \Longrightarrow M. \end{array}$$

On one hand, this example generalizes the LA-groupoid that encodes the prequantization of a symplectic groupoid, see [95]; on the other hand, it provides a geometric construction for general 1-shifted symplectic groupoids that was missing in the literature. Moreover, it was shown in [56, 58] that equivariant gerbes give prequantizations for 1-shifted symplectic groupoids, thus we expect that a connection with curving should induce an isomorphism between the Atiyah algebroid of the equivariant gerbe and (38).

Example 3.30 (Gauge transformations and shifted Lagrangians). Let $(G \Rightarrow M, \omega + H)$ be a Lie groupoid endowed with a closed +1 shifted 2-form as in the preivous example. Now assume that in addition we get a $B \in \Omega^2(M)$ satisfying

$$\omega = \mathbf{s}^* B - \mathbf{t}^* B \quad \text{and} \quad H = dB.$$

The above equations are the ones given in Corollary 3.27. Hence, we get that B gives a gauge transformation between the trivial extension and the prequantum bundle of (ω, H) as defined in Example 3.29.

When $(G \Rightarrow M, \omega + H)$ is a 1-shifted symplectic groupoid, an example of this situation is given by their *Hamiltonian spaces*, see e.g. [96]. Indeed, their quantization was considered in [56] using relative equivariant S^1 -gerbes.

Example 3.31 (Quasi Poisson groupoids). We already saw in Proposition 3.22 that the cotangent of a quasi Poisson groupoid ($G \Rightarrow M, \Pi, \pi$) is a quasi LA-groupoid. Now we can reinterpret

$$\Pi \in C^2_{mul}(T^*[1]G) \text{ and } \pi \in C^3(A^*[1])$$

in the light of Proposition 3.26 and note that the quasi Poisson equations give us exactly the equations in (37). Then we get that the cotangent bundle of a quasi Poisson groupoid has a central extension that is again a quasi LA-groupoid

This gives us the higher analogue of the prequantum bundle studied in [49] in the case of Poisson manifolds.

3.6.3. L_2 -algebra actions on Lie groupoids. Recall that a L_2 -algebra is a tuple⁴ ($\mathfrak{h} \xrightarrow{\partial} \mathfrak{g}, [\cdot, \cdot], K$) as in (20) satisfying (22)-(25). In this case, Theorem 3.15 recovers the result in [9] which shows that ($\mathfrak{h} \to \mathfrak{g}, \partial, [\cdot, \cdot], K$) is equivalent to the quasi LA-groupoid (($\mathfrak{h} \oplus \mathfrak{g} \Rightarrow \mathfrak{g}; * \Rightarrow *$), $Q_{CE}, q = K$).

Definition 3.32. An infinitesimal action of an L_2 -algebra $(\mathfrak{h} \xrightarrow{\partial} \mathfrak{g}, [\cdot, \cdot], K)$ on a Lie groupoid $G \rightrightarrows M$ is an L_{∞} -morphism $\Psi : (\mathfrak{h} \xrightarrow{\partial} \mathfrak{g}, [\cdot, \cdot], K) \rightarrow (\Gamma A \rightarrow \mathfrak{X}_{mul}(G), d, [\cdot, \cdot]).$

A characterization of L_{∞} -algebra actions on graded manifolds was given in [75, Theorem 4.4]. Along the same lines, we can give the following characterization.

Proposition 3.33. Let $(\mathfrak{h} \xrightarrow{\partial} \mathfrak{g}, [\cdot, \cdot], K)$ be a L_2 -algebra with associated quasi LA-groupoid $((\mathfrak{h} \oplus \mathfrak{g} \rightrightarrows \mathfrak{g}; \ast \rightrightarrows \ast), Q_{CE}, q_{CE} = K)$. There is a one to one correspondence between:

(1) Infinitesimal actions of a L_2 -algebra $(\mathfrak{h} \xrightarrow{\partial} \mathfrak{g}, [\cdot, \cdot], K)$ on a Lie groupoid $G \rightrightarrows M$.

⁴For simplicity we consider ∇ as part of $[\cdot, \cdot]$.

(2) Quasi LA-groupoid structures on $(\mathfrak{h} \oplus \mathfrak{g}) \times G \rightrightarrows \mathfrak{g} \times M$ for which the projection to $((\mathfrak{h} \oplus \mathfrak{g} \rightrightarrows \mathfrak{g}; \ast \rightrightarrows \ast), Q_{CE}, q_{CE})$ is a quasi LA-groupoid morphism.

On the one hand, the above actions are a global counterpart of the actions considered in [33, 75]. On the other hand, they are the infinitesimal counterpart of the Lie 2-group actions on differentiable stacks considered in [28, 48].

Remark 3.34. The above Proposition allows us to define infinitesimal actions of quasi LAgroupoids on Lie groupoids by considering projectable quasi LA-groupoids on the pullback of a groupoid morphism.

Consider an *n*-graded Lie groupoid $\mathcal{G} \rightrightarrows \mathcal{M}$ with *n*-graded Lie algebroid $\mathcal{A} \rightarrow \mathcal{M}$. Then $(\Gamma^{\bullet}\mathcal{A} \rightarrow \mathfrak{X}^{1,\bullet}_{mul}(\mathcal{G}), [\cdot, \cdot], d)$ is a \mathbb{Z} -graded L_2 -algebra as in [16, §2.1]. Therefore, one can extend Definition 3.32 to graded groupoids by considering actions of \mathbb{Z} -graded L_2 -algebra as L_{∞} -morphisms that preserves the \mathbb{Z} -degree. In particular, for the \mathbb{Z} -graded Lie algebra ($\mathbb{R}[-1]$) we get the following result.

Proposition 3.35. Let $\mathcal{G} \rightrightarrows \mathcal{M}$ be an *n*-graded Lie groupoid. There is a one to one correspondence between:

- Infinitesimal actions of the \mathbb{Z} -graded Lie algebra $\mathbb{R}[-1]$ on $\mathcal{G} \rightrightarrows \mathcal{M}$.
- Quasi Q-groupoid structures on $\mathcal{G} \rightrightarrows \mathcal{M}$.

Proof. First notice that $\wedge^2 \mathbb{R}[-1] = \operatorname{Sym}^2 \mathbb{R} = \mathbb{R}$. Therefore, an infinitesimal action of the \mathbb{Z} -graded Lie algebra $\mathbb{R}[-1]$ on $\mathcal{G} \rightrightarrows \mathcal{M}$ is the same as two linear maps

$$\Psi: \mathbb{R} o \mathfrak{X}^{1,1}_{mul}(\mathcal{G}) \quad ext{and} \quad K: \mathbb{R} o \Gamma^2 \mathcal{A}$$

satisfying

$$0 = \frac{1}{2}[\Psi(1), \Psi(1)] + dK(1) \text{ and } 0 = [K(1), \Psi(1)].$$

Hence, $Q = \Psi(1)$ and q = K(1) is a quasi Q-structure.

This observation extends the characterization of Q-manifolds as actions of $\mathbb{R}[-1]$ to the world of quasi Q-manifolds.

3.6.4. 2-term representations up to homotopy. Let $A \to M$ be a Lie algebroid and let $\mathbf{V} := V_0 \xrightarrow{\partial} V_1$ be a 2-term complex of vector bundles over M. A representation up to homotopy of A on \mathbf{V} consists of a pair of A-connections ∇^i on V_i for i = 0, 1 and a 2-form $\omega \in \Omega^2(A, \hom(V_1, V_0))$ such that

$$\partial \circ \nabla^1 = \nabla^0 \circ \partial, \quad \partial \circ \omega = \omega_1, \quad \omega \circ \partial = \omega_0, \quad \nabla \omega = 0;$$

where ω_i is the curvature of ∇^i and ∇ is the connection determined by ∇^0 and ∇^1 on $\Omega^{\bullet}(A, \hom(V_1, V_0))$.

It was pointed out in [5, 74] that such a representation up to homotopy can be described as a morphism of L_2 -algebroids $A \to \mathfrak{gl}(\mathbf{V})$ where

$$\mathfrak{gl}(\mathbf{V}) := \hom(V_1, V_0) \xrightarrow{(\partial \circ -, -\circ \partial)} \operatorname{Der}(V_1) \times_{TM} \operatorname{Der}(V_0)$$

with the natural two bracket and zero 3-bracket. Thanks to Theorem 3.15, such objects are in one to one correspondence with morphisms between the quasi LA-groupoids



We think that this interpretation makes their integration in terms of 2-functors more natural, see [17].

3.6.5. Atiyah algebroids of 2-bundles and 2-connections. Throughout this section we will use the following notation. Let us denote by $H \rtimes K \rightrightarrows K$ a strict Lie 2-group [10], a principal 2-bundle is a Lie groupoid $\mathbf{P} \rightrightarrows \mathbf{Q}$ equipped with a multiplicative free and proper action of $H \rtimes K \rightrightarrows K$ such that the quotient is a manifold M regarded as a unit groupoid $M \rightrightarrows M$. Such objects were identified in [11] as another avatar of the nonabelian bundle gerbes of [7, 18], see also [78] for alternative definitions.

Let us start by giving some context that will serve as a motivation for our definition. A connection on a principal G-bundle P can be defined globally as a horizontal invariant distribution but it is often more convenient to view them as a splitting of the exact sequence

$$0 \to P \times_G \mathfrak{g} \to \mathbb{A}_P \to TM \to 0,$$

where $\mathbb{A}_P = TP/G$ is the Atiyah algebroid of P, see e.g. [66, §5]. Moreover, this point of view allow us to characterize the flat connections as the splitting that are Lie algebroid morphisms. Now we will show how, following the same approach, we can recover the notion of connection for a 2-bundle. The Atiyah algebroids of \mathbf{P} and \mathbf{Q} fit into an LA-groupoid

$$\mathbb{A}_{\mathbf{P}} \rightrightarrows \mathbb{A}_{\mathbf{O}}$$

Then we have the canonical projections π_i to the tangent of the base manifold constitute a morphism of LA-groupoids. A splitting of those projection maps compatible with the corresponding VB-groupoid structure is given by a single bundle map F_0 as in the following diagram



Definition 3.36. A 2-connection on $\mathbf{P} \rightrightarrows \mathbf{Q}$ is a pair (F, τ_0) , where $\tau_0 : \wedge^2 TM \to C$ is a vector bundle morphism and C is the core of the VB-groupoid $\mathbb{A}_{\mathbf{P}} \rightrightarrows \mathbb{A}_{\mathbf{Q}}$, such that

(40)
$$dF_0 + \frac{1}{2}[F_0, F_0] = \partial \tau_0;$$

where $\partial: C \to \mathbb{A}_{\mathbf{Q}}$ is the differential in the core sequence of $\mathbb{A}_{\mathbf{P}} \rightrightarrows \mathbb{A}_{\mathbf{Q}}$.

Remark 3.37. Equation (40) is called the vanishing of the 'fake curvature' condition [18]. In principle, we could drop such a requirement for τ_0 in the definition of a 2-connection (this is the approach taken in [93] for instance, see Remark 3.39). However, this equation is necessary for being able to define two-dimensional holonomy in a reasonable sense [11, Prop. 2.14], see also [69].

The obstruction for (F, τ) to satisfy the coherence law (11) of a quasi LA-groupoid morphism is what we define as the the 3-form curvature of the 2-connection:

(41)
$$\Omega_{(F,\tau)} := d\tau_0 + [F_0, \tau_0].$$

Proposition 3.38. The maps (F, τ) defined above determine a quasi LA-groupoid morphism

$$(F,\tau): (TM \rightrightarrows TM) \rightarrow (\mathbb{A}_{\mathbf{P}} \rightrightarrows \mathbb{A}_{\mathbf{Q}})$$

if and only if (41) vanishes.

Proof. Since both are VB-groupoids over a unit groupoid and $TM \rightrightarrows TM$ does not have core we can use Theorem 3.15 to conclude that a quasi LA-groupoid morphism is given by F_0 and a vector bundle map $\tau_0 : \wedge^2 TM \rightarrow C$, where C is the core of the VB-groupoid $\mathbb{A}_{\mathbf{P}} \rightrightarrows \mathbb{A}_{\mathbf{Q}}$, satisfying equations (26)-(29). In this case (26) and (28) are automatic, (27) is equivalent to (40) and the vanishing of (41) is exactly (29).

Remark 3.39. It follows from the previous result that a flat 2-connection is exactly a splitting of $\pi = (\pi_0, \pi_1)$ which is a quasi-LA groupoid morphism (equivalently, an L_2 -algebroid morphism). A flat 2-connection in our sense is also flat as in [93, Def. 5.1.5]. Since we demand multiplicativity in Definition 3.36, our 2-connections are *fake-flat connections* in the sense of [93, Def. 5.1.5]. A closely related result is [47, Thm. 5.43] where the authors work at the level of the Lie 2-algebras of multiplicative sections rather than with the LA-groupoids themselves.

Let us briefly indicate how our approach gives a unified conceptual interpretation for the definitions in [8, 11, 13].

Example 3.40 (Connections on principal 2-bundles [8, 11]). Rephrasing our discussion at the beginning of this subsection, let $H \rtimes K \rightrightarrows K$ be a strict Lie 2-group and consider the trivial principal 2-bundle over a manifold M regarded as a unit groupoid. Then a 2-connection in our sense on the corresponding Atiyah algebroid recovers exactly the definition of a 2-connection with vanishing fake curvature (40).

One can also consider principal 2-bundles whose base space is an arbitrary groupoid $G \Rightarrow M$ instead of just the unit groupoid $M \Rightarrow M$ of a manifold. Those objects are called *principal bundle groupoids* in [40] (see also [47]), we use the same notation as before. Due to Proposition 3.38 we can extend Definition 3.36 to principal bundle groupoids by saying that a *flat 2-connection* on a principal bundle groupoid $\mathbf{P} \Rightarrow \mathbf{Q}$ is a quasi LA-groupoid morphism

$$(F,\tau): (TG \rightrightarrows TM) \to (\mathbb{A}_{\mathbf{P}} \rightrightarrows \mathbb{A}_{\mathbf{Q}})$$

that is an splitting for the anchor of the Atiyah algebroid. If (F, τ) does not satisfy the coherence law (11), we say that (F, τ) is a 2-connection.

A relevant example of this more general framework is the following one.

Example 3.41 (Connective structures on gerbes over differentiable stacks [13]). An S^1 -central extension of a Lie groupoid $G \rightrightarrows M$ can be viewed as a Lie groupoid $P \rightrightarrows M$ equipped with a free multiplicative S^1 -action:

where π is the quotient map. Hence, $P \rightrightarrows M$ gives a principal bundle groupoid.

A connection and a curving on such a principal bundle [13, Def. 4.20] is a form $\theta \in \Omega^1(P)$ which is a usual principal connection together with $B \in \Omega^2(M)$ such that

$$\delta \theta = 0$$
 and $d\theta = \delta B$.

The 3-curvature of the connective structure (θ, B) is $H := dB \in \Omega^3(M)$. Following Example 3.29, one shows that the Atiyah algebroid of P is the \mathbb{R} -central extension of $TG \rightrightarrows TM$ defined by the closed 1-shifted 2-form $\omega + H$ where $\omega = d\theta \in \Omega^2(P)$. From the discussion in Example 3.30 we get that $(F, \tau) := ((\mathrm{id}_{TP}, -\theta), 0)$ is a 2-connection.

Remark 3.42. A principal 2-bundle $\mathbf{P} \rightrightarrows \mathbf{Q}$ over M with respect to the Lie 2-group action of $H \rtimes K \rightrightarrows K$ can be regarded as a generalized morphism from the 2-stack represented by M to the 2-stack represented by $H \rtimes K \rightrightarrows K$ [41]. However, it is convenient to also consider the 2-stacks represented by double Lie groupoids as possible targets for such generalized morphisms. This generalization is given by considering Lie groupoids which are principal bundles for double Lie groupoid actions instead of just Lie 2-group actions; we get in this way a more general family of multiplicative bundles, see [89, §3.3] for the definition of such multiplicative (or morphic) actions. We hope to extend our description of (flat) 2-connections to that setting in future work.

4. Degree two quasi Q-groupoids: quasi CA-groupoids

4.1. Roytenberg-Severa correspondence. In this subsection we summarize some well known facts about degree 2 manifolds that we will use later. Some standard references are [60, 84, 85, 87].

The most important examples of degree 2 manifolds are arguably the ones induced by Courant algebroids. Recall that a *Courant algebroid* is a vector bundle $E \to M$ equipped with a non-degenerate symmetric bilinear form $\langle \cdot, \cdot \rangle$, a bundle map $\rho : E \to TM$, and a bracket $[\![\cdot, \cdot]\!]$ such that

(1) $\llbracket e_1, fe_2 \rrbracket = \rho(e_1)(f)e_2 + f\llbracket e_1, e_2 \rrbracket$,

- $(2) \ \rho(e_1)(\langle e_2, e_3 \rangle) = \langle \llbracket e_1, e_2 \rrbracket, e_3 \rangle + \langle e_2, \llbracket e_1, e_3 \rrbracket \rangle,$
- $(3) \quad \llbracket e_1, \llbracket e_2, e_3 \rrbracket \rrbracket = \llbracket \llbracket e_1, e_2 \rrbracket, e_3 \rrbracket + \llbracket e_2, \llbracket e_1, e_3 \rrbracket \rrbracket,$
- (4) $[\![e_1, e_2]\!] + [\![e_2, e_1]\!] = \mathcal{D}\langle e_1, e_2 \rangle,$

for all $f \in C^{\infty}(M)$ and $e_i \in \Gamma(E)$, where $\mathcal{D}: C^{\infty}(M) \to \Gamma(E)$ is defined by

 $\langle \mathcal{D}f, e \rangle = \rho(e)(f).$

With this definition we can state [85, Theorem 4.5] as follows.

Proposition 4.1. There is a one to one correspondence between:

- (1) Vector bundles $E \to M$ equipped with a nondegenerate symmetric bilinear form $\langle \cdot, \cdot \rangle$ and degree 2 symplectic manifolds.
- (2) Courant algebroids and degree 2 symplectic Q-manifolds.

The above result was extended to the multiplicative case in [60, 72, 79] through the following definitions.

A VB-groupoid $(H \rightrightarrows E; G \rightrightarrows M)$ has a multiplicative pairing if the vector bundle $H \rightarrow G$ is equipped with a nondegenerate symmetric bilinear form $\langle \cdot, \cdot \rangle$ and $\operatorname{Gr}(\mathfrak{m}_H) \rightarrow \operatorname{Gr}(\mathfrak{m}_G)$ is a lagrangian subbundle of $\overline{H} \times H \times H \rightarrow G \times G \times G$ where the $\overline{\cdot}$ denotes minus the pairing on that factor and the inclusion $\operatorname{Gr}(\mathfrak{m}_G) \hookrightarrow G \times G \times G$ is given by $(g, h) \mapsto (\mathfrak{m}_G(g, h), g, h)$ for all $g, h \in G$ composable.

Definition 4.2. A VB-groupoid $(H \rightrightarrows E; G \rightrightarrows M)$ is a *CA-groupoid* if $(H \rightarrow G, \langle \cdot, \cdot \rangle, \llbracket \cdot, \cdot \rrbracket, \rho)$ is a Courant algebroid such that $\operatorname{Gr}(\mathfrak{m}_H) \rightarrow \operatorname{Gr}(\mathfrak{m}_G)$ is a Dirac structure of $\overline{H} \times H \times H \rightarrow G \times G \times G$, see [59, Def. 2.1].

Proposition 4.3. There is a one-to-one canonical correspondence between:

- (1) VB-groupoids endowed with a multiplicative pairing and degree 2 symplectic groupoids.
- (2) CA-groupoids and degree 2 symplectic Q-groupoids.

4.2. **Degree 2 symplectic quasi Q-groupoids.** Let $((H \Rightarrow E; G \Rightarrow M), \langle \cdot, \cdot \rangle)$ be a VB-groupoid endowed with a multiplicative pairing and denote by $(\mathcal{G} \Rightarrow \mathcal{M}, \{\cdot, \cdot\})$ the corresponding degree 2 symplectic groupoid. Motivated by Proposition 4.3 we can introduce the following definition.

Definition 4.4. We say that $((H \Rightarrow E; G \Rightarrow M), \langle \cdot, \cdot \rangle)$ carries a quasi *CA*-groupoid structure if $(\mathcal{G} \Rightarrow \mathcal{M}, \{\cdot, \cdot\})$ is a degree 2 symplectic quasi *Q*-groupoid for some $\Theta \in C^3_{mul}(\mathcal{G})$ and $\theta \in C^4(\mathcal{M})$.

As in the case of quasi LA-groupoids we will not give a classical description of them. Instead we will give several examples that motivate our definition.

Based on [61], it is natural to ask how to characterize the doubles of the categorifed Lie bialgebroids described in §3.5. Using Definition 4.4 together with Proposition 2.21 we get a satisfactory answer that we summarize in the next example.

Example 4.5 (Doubles of categorified Lie bialgebroids). Let $(H \Rightarrow E; G \Rightarrow M)$ be a VB-groupoid with core C and denote by $H[1] \Rightarrow E[1]$ the associated 1-graded Lie groupoid. As explained in §3.5, a categorified Lie bialgebroid structure (in its most general form) corresponds to a Maurer-Cartan element

$$\Psi = ((f, F), (q, Q), t, (\Pi, \pi), (\Phi, \phi)) \in \mathcal{V}^{1}_{H[1]}.$$

Then Proposition 2.21 implies that $(T^*[2]H[1] \rightrightarrows \mathcal{A}^*_{H[1]}[2], \{\cdot, \cdot\}_{can})$ with

$$\Theta = F + Q + \Pi + \Phi \in C^3_{mul}(T^*[2]H[1]) \quad \text{and} \quad \theta = f + q + t + \pi + \phi \in C^4(\mathcal{A}^*_{H[1]}[2])$$

is a degree 2 symplectic quasi Q-groupoid. It is well known, see e.g. [79], that the degree 2 symplectic groupoid $(T^*[2]H[1] \Rightarrow \mathcal{A}^*_{H[1]}[2], \{\cdot, \cdot\}_{can})$ corresponds to the VB-groupoid $(H \oplus H^* \Rightarrow E \oplus C^*; G \Rightarrow M)$ and the canonical pairing is multiplicative. Combining this fact with our results we get that this VB-groupoid carries a quasi CA-groupoid structure. When $\theta = 0$ we get a genuine CA-groupoid, for an extensive study of this case and also of its infinitesimal counterpart see [68].

An interesting consequence of the above example is that we can give a more conceptual proof of Theorem 3.6; in fact, this is a generalization of that result.

Proposition 4.6. Let $(H \rightrightarrows E; G \rightrightarrows M)$ be a VB-groupoid and denote by $H[1] \rightrightarrows E[1]$ the associated 1-graded Lie groupoid. Then

$$(T^*[2]H[1] \rightrightarrows \mathcal{A}^*_{H[1]}[2], \{\cdot, \cdot\}_{can}) \quad and \quad (T^*[2]H^*[1] \rightrightarrows \mathcal{A}^*_{H^*[1]}[2], \{\cdot, \cdot\}_{can})$$

are symplectomorphic groupoids. Moreover, the isomorphism sends

$$\Psi = \left((f, F), (q, Q), t, (\Pi, \pi), (\Phi, \phi) \right) \rightsquigarrow \Psi' = \left((\phi, \Phi), (\pi, \Pi), t, (Q, q), (F, f) \right).$$

Proof. This is a straightforward graded version of [66, Thm. 11.5.14].

In §3.6 we already saw some examples of categorified Lie bialgebroids. Therefore Example 4.5 produces the doubles for these objects. Another relevant example is the following.

Example 4.7. Let $G \rightrightarrows M$ be a groupoid and consider the Q-groupoid $(T[1]G \rightrightarrows T[1]M, Q = d_{dr})$. A categorified Lie bialgebroid given by the Maurer-Cartan element

$$\Psi = \left((f, F), (0, Q), 0, (0, 0), (0, 0) \right) \in \mathcal{V}_{T[1]G}^1$$

is the same as a closed 1-shifted 3-form on the groupoid, i.e. $F\in \Omega^3(G)$ and $f\in \Omega^4(M)$ such that

$$\mathbf{m}^*F = \mathrm{pr}_1^*F + \mathrm{pr}_2^*F, \quad dF = \mathbf{s}^*f - \mathbf{t}^*f \quad \text{and} \quad df = 0$$

The corresponding double is a quasi CA-groupoid on

$$TG \oplus T^*G \rightrightarrows TG \oplus A^*$$

with the canonical pairing and anchor, Courant-Dorfman bracket twisted by the 3-form F and twisting 4-form $\mathbf{s}^* f - \mathbf{t}^* f$, as in [45].

Following the work of one of us, one can upgrade the previous example to show that all the exact twisted CA-groupoids give examples of quasi CA-groupoids.

Example 4.8 (Exact twisted CA-groupoids). Let $G \rightrightarrows M$ be a Lie groupoid and pick a closed 2-shifted 2-form on it, i.e. an $\omega \in \Omega^2(G^{(2)})$, a $F \in \Omega^3(G)$ and a $f \in \Omega^4(M)$ satisfying

$$\delta \omega = 0, \quad d\omega = -\delta F, \quad dF = \delta f, \quad df = 0.$$

Then we can use ω to twist the canonical VB-groupoid structure on $TG \oplus T^*G \rightrightarrows TM \oplus A^*$ by defining the new multiplication \mathfrak{m}_{ω} as follows:

$$\operatorname{Graph}(\mathfrak{m}_{\omega}) := \left\{ (X \oplus \alpha, Y \oplus \beta, Z \oplus \gamma) \in \overline{\mathbb{T}G} \times \mathbb{T}G \times \mathbb{T}G \middle| \begin{array}{l} \mathfrak{m}_{G}^{*} \alpha = \mathrm{pr}_{1}^{*}\beta + \mathrm{pr}_{2}^{*}\gamma + i_{(Y,Z)}\omega \\ X = T\mathfrak{m}(Y,Z), \end{array} \right\}.$$

Now the Courant-Dorfman on $\mathbb{T}G$ can be twisted by F to produce a twisted CA-groupoid structure with the 4-form twist given by f. It was shown in [2] that all the exact twisted CAgroupoids are of this form. The function $\Theta = F + d_{dr} \in C^3(T^*[2]T[1]G)$ is multiplicative with respect to the multiplication \mathfrak{m}_{ω} and, since $\theta = f \in C^4(T[1]M) \subset C^4(T[1]A^*[1])$, the following equations hold:

$$\{\Theta,\Theta\}_{can} = \mathbf{s}^*\boldsymbol{\theta} - \mathbf{t}^*\boldsymbol{\theta} = \mathbf{s}^*_{\omega}\boldsymbol{\theta} - \mathbf{t}^*_{\omega}\boldsymbol{\theta} \quad \text{and} \quad \{\Theta,\mathbf{t}^*_{\omega}\boldsymbol{\theta}\}_{can} = \{\Theta,\mathbf{t}^*\boldsymbol{\theta}\}_{can} = 0$$

because $\mathbf{s}_{\omega}^*|_{C^4(T[1]M)} = \mathbf{s}^*|_{C^4(T[1]M)}$ and the same for the target. Therefore all these exact twisted CA-groupoids produce examples of quasi CA-groupoids.

Example 4.9 (Central extensions). Let $(\mathcal{G} \rightrightarrows \mathcal{M}, \{\cdot, \cdot\}, \Theta, \theta)$ be a degree 2 symplectic quasi Q-groupoid. Then one can construct an $\mathbb{R}[2]$ central extension, $\mathcal{G} \times \mathbb{R}[2] \rightrightarrows \mathcal{M}$, that is a degree 2 quasi Q-groupoid with

$$\widehat{Q}(f) = Q(f), \quad \widehat{Q}(\lambda) = \Theta, \quad \widehat{q}^r(f) = q^r(f) \text{ and } \widehat{q}^r(\lambda) = \mathbf{t}^* \theta,$$

where $f \in C^{\bullet}(\mathcal{G})$ and $\lambda \in C^{2}(\mathbb{R}[2])$ is the generator. The interested reader can verify that this is indeed a quasi Q-groupoid by following the same approach outlined in the proof of Proposition 3.26.

HOMOLOGICAL VECTOR FIELDS OVER DIFFERENTIABLE STACKS

5. Epilogue: Lie and Courant algebroids through the stacky glasses

In the course of this work we have described several structures on (graded) Lie groupoids and we have shown that they can be transported along (and compared using) Morita equivalences. This allows us to re-interpret our previous results in terms of differentiable stacks.

Recall that a *differentiable stack* is a stack \mathfrak{m} , i.e. a category fibered in groupoids satisfying sheaf-like properties (see [13, Def. 2.10]), that admits a presentation, i.e. a manifold M equipped with a surjective representable submersion $m: M \to \mathfrak{m}$.

It is well known that there exists an equivalence of 2-categories between the 2-category of differentiable stacks and the 2-category of Lie groupoids up to Morita equivalence, see e.g. [82, 13] for the precise mathematical formulation. Roughly speaking the correspondence works as follows. For a given differentiable stack \mathfrak{m} one can take a presentation $m: M \to \mathfrak{m}$ and then show that $M \times_{\mathfrak{m}} M \rightrightarrows M$ is a Lie groupoid. Moreover, different choices of presentation give Morita equivalent groupoids. Conversely, given a Lie groupoid $G \rightrightarrows M$ one shows that the category of G-Principal bundles \mathfrak{m} is a differentiable stack with a presentation given by the quotient morphism $m: M \to \mathfrak{m}$ corresponding to the universal trivial G-principal bundle.

Let $G \rightrightarrows M$ be a Lie groupoid with corresponding differentiable stack \mathfrak{m} . The equivalence described above allows us to define structures on \mathfrak{m} using a Lie groupoid representative $G \rightrightarrows M$. For example, it is shown in [46, 80] that vector fields on \mathfrak{m} are modeled by the dgla

$$L_G^{\bullet} = (\Gamma A \to \mathfrak{X}_{mul}(G), d, [\cdot, \cdot]).$$

Recently, that work was extended to 1-shifted multivector fields on \mathfrak{m} by introducing the dgla $(V_G^{\bullet}, d, [\cdot, \cdot])$ given in (8), see [16].

Recall that a homological vector field on a degree n manifold \mathcal{M} is a Maurer-Cartan element on $(\mathfrak{X}^{1,\bullet}(\mathcal{M}), [\cdot, \cdot])$. In §2 we introduced n-graded differentiable stacks \mathfrak{M} , i.e. differentiable stacks in the category of degree n manifolds, as Morita equivalence classes of n-graded Lie groupoids $\mathcal{G} \rightrightarrows \mathcal{M}$. The results mentioned in the previous paragraph show that vector fields on \mathfrak{M} are modeled by the dgla $(L^{\bullet}_{\mathcal{G}}, d, [\cdot, \cdot])$ (see Proposition 2.12) and thus we can define a homological vector field on \mathfrak{M} as the class of a Maurer-Cartan element in $L^{\bullet}_{\mathcal{G}}$. These Maurer-Cartan elements are, by Proposition 2.13, the same as quasi Q-structures (Definition 2.8); on the other hand, by Proposition 3.35 these are the same as actions of $\mathbb{R}[-1]$ on \mathfrak{M} . Finally, when it comes to Poisson structures, we argue in §2.4 that, by replacing $L^{\bullet}_{\mathcal{G}}$ with $\mathcal{V}^{\bullet}_{\mathcal{G}}$, one gets the notion of a PQ-structure on \mathfrak{M} .

Using the fact that the category of 1-manifolds is equivalent to the category of vector bundles, in §3 we study differentiable stacks in the category of vector bundles. It was explained in [39] (see also [44]) that the VB-Morita equivalence class of a VB-groupoid can be used to define a 2-vector bundle⁵ over a differentiable stack. The two key definitions of §3 are:

- A 1-shifted Lie algebroid structure on a 2-vector bundle over a differentiable stack is the Morita class of a quasi LA-groupoid as introduced in Definition 3.4.
- A linear 1-shifted Poisson structure on a 2-vector bundle over a differentiable stack is the Morita class of a quasi PVB-groupoid as introduced in Definition 3.5.

Theorem 3.6 shows that the 1-dual of a 1-shifted Lie algebroid on a 2-vector bundle over a differentiable stack is a linear 1-shifted Poisson structure. In the light of the above result, §3.5 treats the concept of 1-shifted Lie bialgebroid.

Theorem 3.15 proves that a 1-shifted Lie algebroid on a 2-vector bundle over a manifold is a L_2 -algebroid up to L_{∞} -quasi isomorphism. In the terminology of [98], one can call these

⁵Compare with [94] where descent is used to define vector bundles over differentiable stacks.

objects *stacky Lie algebroids*, we expect that they are the infinitesimal counterpart of the stacky groupoids introduced in [98].

With this language, the examples at the end of §3 acquire the following interpretation:

- (1) In §3.6.1 we show that the (1-shifted) cotangent of a 1-shifted Poisson structure on a differentiable stack admits a canonical 1-shifted Lie bialgebroid structure.
- (2) In §3.6.2 we exhibit the prequantum 1-shifted Lie algebroids of 1-shifted symplectic and Poisson structures on differentiable stacks.
- (3) In §3.6.3 we construct the 1-shifted action Lie algebroid associated to the infinitesimal action of a stacky Lie algebra on a differentiable stack.
- (4) In §3.6.4 we prove that representations up to homotopy of a Lie algebroid A on a 2-term complex \mathbf{V} are in 1-1 correspondence with 1-shifted Lie algebroid morphisms from A to $\mathfrak{gl}(\mathbf{V})$.
- (5) In §3.6.5 we demonstrate that flat 2-connections on a principal 2-bundle over a manifold $\mathfrak{p} \to M$ are the same as 1-shifted Lie algebroid morphisms from TM to the 1-shifted Atiyah algebroid $\mathbb{A}_{\mathfrak{p}}$.

Finally, §4 is devoted to the study of *Courant algebroids over differentiable stacks*. In particular, examples of such objects are given by doubles of 1-shifted Lie bialgebroids as in Example 4.5 and by Exact twisted CA-groupoids as in Example 4.8.

Observe that in our Definition 4.4 of a quasi CA-groupoid we have a genuine symplectic structure. This means that quasi CA-groupoid structures cannot be transported along Morita equivalences. The way of restoring Morita invariance into the picture, and thus defining a Courant algebroid over a differentiable stack, is by considering a shifted symplectic structure in the following way which is completely analogous to the situation of quasi-symplectic groupoids [25, 96]. A shifted Courant algebroid over a differentiable stack is the Morita class of a degree 2 quasi Q-groupoid ($\mathcal{G} \Rightarrow \mathcal{M}, Q, q$) together with a 1-shifted 2-form of degree 2, i.e. a pair of differential forms $\omega_1 \in \Omega^{2,2}(\mathcal{G})$ and $\omega_0 \in \Omega^{3,2}(\mathcal{M})$, satisfying the following properties:

- (1) closedness: $\mathfrak{m}^*\omega_1 = \operatorname{pr}_1^*\omega_1 + \operatorname{pr}_2^*\omega_1$, $d\omega_1 = \mathfrak{s}^*\omega_0 \mathfrak{t}^*\omega_0$ and $d\omega_0 = 0$;
- (2) non-degeneracy: the form ω_1 is homologically non-degenerate by inducing a quasiisomorphism between the tangent $\mathcal{A} \xrightarrow{\rho} T\mathcal{M}$ and cotangent $T^*[2]\mathcal{M} \xrightarrow{\rho^*} \mathcal{A}^*[2]$ complex of the graded Lie groupoid.
- (3) (q, Q) invariance: $\mathcal{L}_Q \omega_1 = 0$, $\mathcal{L}_Q \mathsf{t}^* \omega_0 = 0$ and $\mathcal{L}_{q^r} \omega_1 = 0$.

Note that, since ω_1 is not closed, we cannot conclude that Q is a hamiltonian vector field for some degree 3 function on \mathcal{G} . Another observation is that, in this situation, \mathcal{A} can be viewed as a "Dirac structure" in $T\mathcal{M} \oplus T^*[2]\mathcal{M}$ with bracket twisted by the 3-form ω_0 , see [92] for the graded version of Courant algebroids.

APPENDIX A. DOUBLE QUASI POISSON GROUPOIDS

Following [19, 63] we say that a *double topological groupoid* (D; H, G; M) is a groupoid object in the category of topological groupoids and is represented by a diagram of the form



A double Lie groupoid is a double topological groupoid as in the previous diagram such that each of the side groupoids is a Lie groupoid and the double source map $(\mathbf{s}^H, \mathbf{s}^G) : D \to H \times_M G$ is a submersion.

Example A.1. Let $G \rightrightarrows M$ be a Lie groupoid. Then the pair groupoid $G \times G$ is a double Lie groupoid with sides $M \times M$ and G over M.

Example A.2. A crossed module of Lie groups (G, H, ∂, α) , see e.g. [10, § 8.4], gives rise to a double Lie groupoid with $D = H \times G$ with sides G and a point. Here D and G are also Lie groups, thus crossed modules can be equivalently described as a group object in the category of Lie groupoids. We refer to them as strict Lie 2-groups.

It is straightforward to see that the Lie functor establishes an equivalence of categories between the categories of 1-connected strict Lie 2-groups and strict Lie 2-algebras. As a (wellknown) consequence of Theorem 3.15, we get that strict Lie 2-algebras are equivalent to *crossed modules of Lie algebras*.

If (D; H, G; M) is a double Lie groupoid, we denote by A_D^h the Lie algebroid of $D \rightrightarrows H$ and by A_D^v the Lie algebroid of $D \rightrightarrows G$. By functoriality, $(A_D^h \rightrightarrows A_G; H \rightrightarrows M)$ and $(A_D^v \rightrightarrows A_H; G \rightrightarrows M)$ are LA-groupoids, see [63] for details. For a given $X \in \Gamma(\wedge^{\bullet} A_D^h)$, its horizontal left and right invariant extensions are denoted by $X^{\mathcal{L}}$ and by $X^{\mathcal{R}}$; while for $Y \in \Gamma(\wedge^{\bullet} A_D^v)$, its vertical left and right invariant extensions are denoted by Y^L and by Y^R .

Definition A.3. A double quasi Poisson groupoid $((D; H, G; M), \Pi, \Psi)$ is a double Lie groupoid (D; H, G; M) which is equipped with a bivector field $\Pi \in \mathfrak{X}^2(D)$ and $\Psi \in \Gamma(\wedge^3 A_D^h)$ satisfying:

- Π is multiplicative over both H and G,
- $[\Pi,\Pi] = \Psi^{\mathcal{L}} \Psi^{\mathcal{R}}$
- Ψ is multiplicative with respect to the VB-groupoid $(\bigoplus^3 A_D^h \Rightarrow \bigoplus^3 A_G; H \Rightarrow M)$: in other words, the contraction map

$$\overline{\Psi}: \left(\bigoplus^{3} (A_{D}^{h})^{*} \rightrightarrows \bigoplus^{3} C^{*}; H \rightrightarrows M \right) \to \mathbb{R}$$

is a groupoid morphism, where C is the core of the VB-groupoid $(A_D^h \rightrightarrows A_G; H \rightrightarrows M)$.

Remark A.4. Notice that the fact that Ψ is multiplicative implies immediately that $\Psi^{\mathcal{L}} - \Psi^{\mathcal{R}}$ is also vertically multiplicative.

Example A.5. A double quasi Poisson groupoid $((D; H, G; M), \Pi, \Psi)$ in which $D \rightrightarrows G$ and $H \rightrightarrows M$ are vector bundles and Π and Ψ are linear is a quasi PVB-groupoid as in Definition 3.5.

Example A.6. A double quasi Poisson groupoid $((D; H, G; M), \Pi, 0)$ in which Π is nondegenerate is a *double symplectic groupoid* [62].

Example A.7. Strict Lie 2-groups as in Example A.2 can be equipped with double quasi Poisson structures in two different ways.

- (1) A double quasi-Poisson groupoid of the form $((D; G, *; *), \Pi, \Psi)$ is a quasi Poisson Lie 2-group in the sense of [35]; if $\Psi = 0$, we get a Poisson 2-group.
- (2) A double quasi-Poisson groupoid of the form $((D; *, G; *), \Pi, \Psi)$ can be called a *vertical* quasi Poisson 2-group in order to distinguish this notion from the one described in the previous example. Notice that a Poisson 2-group is also a vertical quasi Poisson 2-group with respect to its group structure.

Many other examples of double quasi Poisson groupoids arise when considering moduli spaces of representations of fundamental groupoids, see [3].

A.1. Quasi Poisson LA-groupoids and multiplicative Lie quasi-bialgebroids. Differentiating vertically a double quasi Poisson groupoid we obtain a quasi Poisson LA-groupoid as in § 3.5.

Proposition A.8. The vertical LA-groupoid of a double quasi Poisson groupoid inherits a canonical quasi Poisson LA-groupoid structure.

Proof. Let $((D; H, G; M), \Pi, \Psi)$ be a double quasi Poisson groupoid. Since Π is multiplicative over G, the formula

$$\delta(X)^R = [\Pi, X^R]$$
 for all $X \in \Gamma(\wedge^{\bullet} A_D^v)$

defines a degree 1 multiplicative derivation $\delta : \Gamma(\wedge^{\bullet} A_D^v) \to \Gamma(\wedge^{\bullet+1} A_D^v)$ of the Schouten bracket on $\Gamma(\wedge^{\bullet} A_D^v)$ determined by its LA-groupoid structure [22].

Note that a quasi Poisson LA-groupoid can be equivalently described as a VB-groupoid $(A_D^v \rightrightarrows A_H; G \rightrightarrows M)$ endowed with an LA-groupoid structure $((A_D^v \rightrightarrows A_H; G \rightrightarrows M), \mathbf{a}_H, [,])$ and a quasi LA-groupoid structure $(((A_D^v)^* \rightrightarrows C^*; G \rightrightarrows M), \delta, \psi)$ such that δ is a derivation of [,]. So it only remains to show that δ can be extended to a quasi LA-groupoid structure.

Note that the linear bivector field determined by δ as in Proposition 3.1 is a quasi Poisson bivector with respect to the linear section in $\wedge^{3}\text{Lie}(A_{D}^{v})$ determined by $\text{Lie}(\Psi)$. Therefore, Theorem 3.6 implies the result.

Remark A.9. Note that the horizontal LA-groupoid of a double quasi Poisson groupoid is automatically a multiplicative Lie quasi-bialgebroid. This means that quasi Poisson LA-groupoids and multiplicative Lie quasi-bialgebroids are in duality as LA-groupoids.

A.2. Integration of quasi Poisson LA-groupoids over a unit groupoid. Theorem A.8 allows to get a quasi Poisson LA-groupoid out of a double quasi Poisson groupoid; the inverse operation involves the integration problem for LA-groupoids which is still an open question, see [89]. For a certain class of quasi Poisson LA-groupoids however we can obtain an integration result which generalizes [35, Thm. 3.2].

Theorem A.10. A quasi Poisson LA-groupoid over a unit groupoid which consists of integrable Lie algebroids is integrable by a double quasi Poisson groupoid.

This result follows from a more fundamental observation which is itself an easy consequence of [76, Corollary 5.4].

Proposition A.11. An LA-groupoid over a groupoid consisting only of units is integrable by a double Lie groupoid if its Lie algebroid of arrows is integrable.

Proof. Let $(A \Rightarrow E; M \Rightarrow M)$ be an LA-groupoid and suppose that A is integrable, then so are E and the core C since they are Lie subalgebroids of A. We have that A is isomorphic to the semidirect product of Lie algebroids $C \rtimes E$. Let H be the source-simply-connected integration of E and let K be the source-simply-connected integration of C. Then the action of E on C integrates to a Lie groupoid action of H on K along the identity on M and the associated semidirect product groupoid $D := K \rtimes H \Rightarrow M$ integrates A [76, Corollary 5.4]. So we have that the action groupoid structure $D \Rightarrow H$ constitutes a double Lie groupoid (D; H, M; M) integrating $(A \Rightarrow E; M \Rightarrow M)$.

Proof of Theorem A.10. This follows from observing that the double Lie groupoid produced by Proposition A.11 is both horizontally and vertically source-simply-connected so one can apply to it the standard integration result of [50]. \Box

Appendix B. Infinitesimal derivations and linear sections

Here we include the proof of a technical result that we need to give a classical proof of Theorem 3.6.

Proposition B.1. Let $(H \rightrightarrows E; G \rightrightarrows M)$ be a VB-groupoid with core C. Denote by $H[1] \rightrightarrows E[1]$ the associated 1-graded groupoid and pick $q \in \Gamma^2 \mathcal{A}_{H[1]} = \Gamma^2(A_H[1])$. Then, there is a linear section π of $\wedge^3 A_{H^*} \rightarrow C^*$ such that

(42)
$$q^{r}(\omega)^{\uparrow} = [\pi^{r}, \omega^{\uparrow}] \qquad \forall \omega \in \Gamma \wedge^{\bullet} H^{*}.$$

Conversely, a linear section π of $\wedge^3 A_{H^*} \to C^*$ induces a right-invariant derivation on $H[1] \Longrightarrow E[1]$ by means of formula (42).

Proof. Let $q \in \Gamma^2 \mathcal{A}_{H[1]}$ and consider its right invariant extension $q^r \in \mathfrak{X}^{1,2}(H[1])$. Proposition 3.1 tells us that there is a corresponding linear $\overline{\pi} \in \mathfrak{X}^3(H^*)$. We have to prove that $\overline{\pi}$ factorizes as in the following diagram

(43)
$$\begin{array}{c} \bigoplus^{3} T^{*}H^{*} \xrightarrow{\overline{\pi}} \mathbb{R} \\ \oplus^{3} t_{T^{*}H^{*}} \downarrow & \pi \\ \oplus^{3} A_{H^{*}}^{*} \end{array}$$

for some linear section π in $\wedge^3 A_{H^*} \to C^*$.

Step 1: $\overline{\pi}$ is tangent to the \mathbf{s}_{H^*} -fibers. We shall show that

$$\overline{\pi}(\alpha,\cdot,\cdot) = 0$$
 for all $\alpha \in \ker t_{T^*H^*} = (\ker T \mathbf{s}_{H^*})^{\circ}$

Let $p: H^* \to G$ be the vector bundle projection. By (3) we get that $q^r(\mathbf{s}^*_G(f)) = 0$ for any $f \in C^{\infty}(M)$. Thus using the definition of $\overline{\pi}$ we have

(44)
$$\overline{\pi}(dp^*\mathbf{s}_G^*f,\cdot,\cdot) = [\overline{\pi},(\mathbf{s}_G^*f)^{\uparrow}](\cdot,\cdot) = q^r(\mathbf{s}_G^*f)^{\uparrow}(\cdot,\cdot) = 0$$

Take $\theta \in \Gamma H^*$, $c \in \Gamma C$ and $h_i \in \Gamma H$ for i = 1, 2 and denote their linear functions by ℓ_{h_i} . Linearity of $\overline{\pi}$ implies that $\overline{\pi}(d\ell_{c^l}, d\ell_{h_1}, d\ell_{h_2}) = \ell_{\Phi(c^l, h_1, h_2)}$ for some $\Phi(c^l, h_1, h_2) \in \Gamma H$. Thus, we have that

(45)
$$q^{r}(\theta)^{\uparrow}(c^{l},h_{1},h_{2}) = [\overline{\pi},\theta^{\uparrow}](d\ell_{c^{l}},d\ell_{h_{1}},d\ell_{h_{2}}) = \langle\theta,\Phi(c^{l},h_{1},h_{2})\rangle + \overline{\pi}(dp^{*}\langle\theta,c^{l}\rangle,d\ell_{h_{1}},d\ell_{h_{2}}) + \overline{\pi}(dp^{*}\langle\theta,h_{1}\rangle,d\ell_{h_{2}},d\ell_{c^{l}}) + \overline{\pi}(dp^{*}\langle\theta,h_{2}\rangle,d\ell_{c^{l}},d\ell_{h_{1}}).$$

Recall that $s_{H^*}^* \ell_c = \ell_{c^l}$ and that H fits into the exact sequences

(46)
$$0 \longrightarrow \mathsf{t}_G^* C \longrightarrow H \xrightarrow{(p, \mathbf{s}_H)} \mathbf{s}_G^* E \longrightarrow 0$$

(47)
$$0 \longrightarrow \mathbf{s}_G^* C \longrightarrow H \xrightarrow{(p, \mathbf{t}_H)} \mathbf{t}_G^* E \longrightarrow 0$$

see e.g. [65]. Then we have two possibilities:

DANIEL ÁLVAREZ AND MIQUEL CUECA

- (1) $\theta = \mathbf{s}_{G}^{*} \alpha$ for $\alpha \in \Gamma E^{*}$, using the dual of (46). In this case $\langle \theta, c^{l} \rangle = -\mathbf{s}_{G}^{*} \langle \alpha, \mathbf{t}_{H}(c) \rangle$ and thus the term $\overline{\pi}(dp^{*}\langle \theta, c^{l} \rangle, d\ell_{h_{1}}, d\ell_{h_{2}})$ vanishes for all h_{i} by (44). Let us choose a splitting of the first sequence $\sigma : \mathbf{s}_{G}^{*} E \to H$. So we can assume that h_{i} lies either in ker \mathbf{s}_{H} or in the image of σ ; in the former case $\langle \mathbf{s}_{H}^{*} \alpha, h_{2} \rangle = 0$ and in the latter case we have that $\langle \mathbf{s}_{H}^{*} \alpha, h_{2} \rangle$ is the pullback by \mathbf{s}_{G} of a function on M so the last two terms of (45) vanish again by (44). As a consequence, plugging the identity $T[2]\mathbf{s}_{H^{*}} \circ q^{r} = 0$ in (45) we get that $\Phi(c^{l}, h_{1}, h_{2})$ lies in ker \mathbf{s}_{H} . Hence $\overline{\pi}$ is tangent to the $\mathbf{s}_{H^{*}}$ -fibers.
- (2) θ lies in $\sigma'(\mathbf{t}_G^* E)^{\circ}$, for $\sigma': \mathbf{t}_G^* E \to H$ a splitting of (47), and thus it is \mathbf{s}_{H^*} -related to $\theta_0 \in \Gamma C^*$. In this case $\overline{\pi}(dp^*\langle\theta, c^l\rangle, d\ell_{h_1}, d\ell_{h_2})$ vanish again by (44). If h_i is of the form v^l for some $v \in \Gamma C$, then $\langle\theta, v^l\rangle = \mathbf{s}_G^*\langle\theta_0, v\rangle$ and so the terms $\overline{\pi}(dp^*\langle\theta, h_i\rangle, \cdot, \cdot)$ vanish. If h_i lies in $\sigma'(\mathbf{t}_G^* E)$ then $\langle\theta, h_i\rangle = 0$; in either case we get then that all the terms in the last row of (45) vanish.

Now let us view the contraction $i_{c^l} : \Gamma(\wedge^{\bullet} H^*) \to \Gamma(\wedge^{\bullet-1} H^*)$ as a graded vector field on H[1]. We have that $[q^r, i_{c^l}] = 0$ since right-invariant vector fields commute with left-invariant vector fields. So we get that

$$q^{r}(\theta)(c^{l},\cdot,\cdot) = q^{r}\langle\theta,c^{l}\rangle(\cdot,\cdot) = (q^{r}\circ\mathbf{s}_{G}^{*})\langle\theta_{0},c\rangle(\cdot,\cdot) = 0$$

and hence $\langle \theta, \Phi(c^l, h_1, h_2) \rangle$ vanishes as well. Since we can choose $\sigma'(\mathbf{t}_G^* E)$ to be complementary to ker \mathbf{s}_H , it follows that $\Phi(c^l, h_1, h_2) = 0$ and so we get $\overline{\pi}(d\ell_{c^l}, d\ell_{h_1}, d\ell_{h_2}) = 0$.

Finally, take $f \in C^{\infty}(G)$ arbitrary. Since $[q^r, i_{c^l}] = 0$ we get that

$$\overline{\pi}(dp^*f, d\ell_{c^l}, d\ell_h) = (i_{c^l} \circ q^r)(f)(h) = (q^r \circ i_{c^l})(f)(h) = 0$$

for all $h \in \Gamma H$. Therefore, $\overline{\pi}(d\ell_{c^l}, \cdot, \cdot) = 0$.

Step 2: $\overline{\pi}$ is right-invariant. Since $\overline{\pi}$ is linear it is enough to check it on three linear sections and on one basic and two linear. The right invariance of q^r action on functions of H[1] is equivalent to

$$(q^r,0)(\mathbf{m}^*_{H[1]}\omega) = \mathbf{m}^*_{H[1]}q^r(\omega) \qquad \omega \in C^{\bullet}(H[1]) = \Gamma \wedge^{\bullet} H^*.$$

Take $f \in C^{\infty}(G)$ and take $(h_i, k_i) \in \Gamma H^{(2)}$ for i = 1, 2. By definition of $\overline{\pi}$ we have that

$$\begin{aligned} (48) \quad & \mathfrak{m}_{H^*}^* \overline{\pi}(dp^*f, d\ell_{\mathfrak{m}_H(h_1,k_1)}, d\ell_{\mathfrak{m}_H(h_2,k_2)}) = & (\mathfrak{m}_{H[1]}^*q^r(f))((h_1,k_1), (h_2,k_2)) \\ &= & ((q^r, 0)\mathfrak{m}_{H[1]}^*f)((h_1,k_1), (h_2,k_2)) \\ &= & (\overline{\pi}, 0)(d\mathfrak{m}_G^*f, d\ell_{(h_1,k_1)}, d\ell_{(h_2,k_2)}) \\ &= & (\overline{\pi}, 0)(\mathfrak{m}_{H^*}^*(dp^*f), \mathfrak{m}_{H^*}^*d\ell_{\mathfrak{m}_H(h_1,k_1)}, \mathfrak{m}_{H^*}^*d\ell_{\mathfrak{m}_H(h_2,k_2)}) \end{aligned}$$

because $m_{H^*}^* \ell_{m_H(X_1,Y_1)} = \ell_{(X_1,Y_1)}$.

Take $\theta \in \Gamma H^*$ and $(h_i, k_i) \in \Gamma H^{(2)}$ for i = 1, 2, 3. We have that

(49)
$$\mathbf{m}_{H^*}^*[\overline{\pi},\theta^{\uparrow}](d\ell_{\mathbf{m}_H(h_1,k_1)},d\ell_{\mathbf{m}_H(h_2,k_2)},d\ell_{\mathbf{m}_H(h_3,k_3)}) = \mathbf{m}_{H^*}^*p^*Q$$

where $Q \in C^{\infty}(G)$ is given by the formula

$$\begin{aligned} Q = & \langle \theta, \Phi(\mathbf{m}_{H}(h_{1},k_{1}),\mathbf{m}_{H}(h_{2},k_{2}),\mathbf{m}_{H}(h_{3},k_{3})) \rangle + \overline{\pi}(dp^{*}\langle\theta,\mathbf{m}_{H}(h_{1},k_{1})\rangle,d\ell_{\mathbf{m}_{H}(h_{2},k_{2})},d\ell_{\mathbf{m}_{H}(h_{3},k_{3})}) \\ & + \overline{\pi}(dp^{*}\langle\theta,\mathbf{m}_{H}(h_{3},k_{3})\rangle,dl_{\mathbf{m}_{H}(h_{1},k_{1})},d\ell_{\mathbf{m}_{H}(h_{2},k_{2})}) + \overline{\pi}(dp^{*}\langle\theta,\mathbf{m}_{H}(h_{2},k_{2})\rangle,d\ell_{\mathbf{m}_{H}(h_{3},k_{3})},d\ell_{\mathbf{m}_{H}(h_{1},k_{1})}) \end{aligned}$$

with $\overline{\pi}(d\ell_{\mathfrak{m}_{H}(h_{1},k_{1})}, d\ell_{\mathfrak{m}_{H}(h_{2},k_{2})}, d\ell_{\mathfrak{m}_{H}(h_{3},k_{3})}) = \ell_{\Phi(\mathfrak{m}_{H}(h_{1},k_{1}),\mathfrak{m}_{H}(h_{2},k_{2}),\mathfrak{m}_{H}(h_{3},k_{3}))}$. Thus, the multiplicativity of a^{r} and the definition of $\overline{\pi}$ imply together that

$$(50) \qquad \mathbf{m}_{H^*}^* p^* Q = \mathbf{m}_{H^*}^* [\overline{\pi}, \theta^{\dagger}] (d\ell_{\mathbf{m}_H(h_1, k_1)}, d\ell_{\mathbf{m}_H(h_2, k_2)}, d\ell_{\mathbf{m}_H(h_3, k_3)}) = (\mathbf{m}_{H[1]}^* q^r(\theta)) ((h_1, k_1), (h_2, k_2), (h_3, k_3)) = ((q^r, 0) \mathbf{m}_{H[1]}^* \theta) ((h_1, k_1), (h_2, k_2), (h_3, k_3)) = [(\overline{\pi}, 0), (\mathbf{m}_{H^*}^* \theta)^{\uparrow}] ((d\ell_{h_1}, d\ell_{k_1}), (dl_{h_2}, d\ell_{k_2}), (dl_{h_3}, d\ell_{k_3})) = (p, p)^* \langle \mathbf{m}_{H}^* \theta, (\Phi, 0) ((h_1, k_1), (h_2, k_2), (h_3, k_3)) \rangle + (\overline{\pi}, 0) (dp^* \langle \mathbf{m}_{H}^* \theta, (h_1, k_1) \rangle, dl_{(h_2, k_2)}, dl_{(h_3, k_3)}) + (\overline{\pi}, 0) (dp^* \langle \mathbf{m}_{H}^* \theta, (h_2, k_2) \rangle, dl_{(h_1, k_1)}, dl_{(h_2, k_2)}) + (\overline{\pi}, 0) (dp^* \langle \mathbf{m}_{H}^* \theta, (h_2, k_2) \rangle, dl_{(h_3, k_3)}, dl_{(h_1, k_1)}).$$

So from the specific form of Q and equation (48) it follows that

 $\mathbf{m}_{G}^{*}\langle\theta,\Phi(\mathbf{m}_{H}(h_{1},k_{1}),\mathbf{m}_{H}(h_{2},k_{2}),\mathbf{m}_{H}(h_{3},k_{3}))\rangle = \langle \mathbf{m}_{H}^{*}\theta,(\Phi,0)((h_{1},k_{1}),(h_{2},k_{2}),(h_{3},k_{3}))\rangle,$ hence we get

$$\begin{split} \mathbf{m}_{H^*}^* \overline{\pi} (d\ell_{\mathbf{m}_H(h_1,k_1)}, d\ell_{\mathbf{m}_H(h_2,k_2)}, d\ell_{\mathbf{m}_H(h_3,k_3)})(\mathbf{m}_H^*\theta) &= \mathbf{m}_{H^*}^* \ell_{\Phi(\mathbf{m}_H(h_1,k_1),\mathbf{m}_H(h_2,k_2),\mathbf{m}_H(h_3,k_3))}(\mathbf{m}_H^*\theta) \\ &= \mathbf{m}_G^* \langle \theta, \Phi(\mathbf{m}_H(h_1,k_1), \mathbf{m}_H(h_2,k_2), \mathbf{m}_H(h_3,k_3)) \rangle \\ &= \langle \mathbf{m}_H^*\theta, (\Phi, 0)((h_1,k_1), (h_2,k_2), (h_3,k_3)) \rangle \\ &= (\overline{\pi}, 0)(d\mathbf{m}_{H^*}^*\ell_{(h_1,k_1)}, d\mathbf{m}_{H^*}^*\ell_{(h_2,k_2)}, d\mathbf{m}_{H^*}^*\ell_{(h_3,k_3)})(\mathbf{m}_H^*\theta) \end{split}$$

proving that $\overline{\pi}$ is right invariant. So it is of the form $\overline{\pi} = \pi^r$ for some $\pi \in \Gamma(\wedge^3 A_{H^*})$.

Step 3: π is linear. Let us consider the restriction map

$$r: \left(\bigoplus^{3} T^{*}H^{*} \rightrightarrows \bigoplus^{3} A_{H^{*}}^{*}; G \rightrightarrows M \right) \rightarrow \left(\bigoplus^{3} H \rightrightarrows \bigoplus^{3} E; G \rightrightarrows M \right)$$

defined by $\alpha \mapsto \alpha|_{\ker Tp} \in p^*H$ for all $\alpha \in T^*H^*$. We have that r is the projection associated to a vector bundle structure and the fact that $\overline{\pi}$ is linear means that the contraction $\overline{\pi} : \bigoplus^3 T^*H^* \to \mathbb{R}$ is a vector bundle map with respect to this structure [22, §6] but then so is its restriction to the vector bundle $\bigoplus^3 A^*_{H^*} \to \bigoplus^3 E$ and this means that π is linear. The derivation q^r defined by (42) is right-invariant. Suppose that we have a linear section

The derivation q^r defined by (42) is right-invariant. Suppose that we have a linear section $\pi \in \Gamma(\wedge^3 A_{H^*} \to C^*)$. Then the 3-vector field $\overline{\pi} = \pi^r$ defined by diagram (43) is also linear. Since

$$\overline{\pi}(dp^*\mathbf{s}_G^*f,\cdot,\cdot) = [\pi^r, p^*f](\cdot,\cdot),$$

we have that $q^r(\mathbf{s}_G^*f) = 0$. On the other hand, the equation $\overline{\pi}(d\ell_{c^l}, \cdot, \cdot) = 0$ applied to (45) implies that $q^r \circ \mathbf{s}_H^* \alpha = 0$ for all $\alpha \in \Gamma E^*$. As a consequence, $T[2]\mathbf{s}_{H[1]} \circ q^r = 0$. Since π^r is right-invariant, we just need to use equations (48) and (50) starting from the equalities in the middle to their endpoints, as this follows the same approach that we used before, the details are left to the reader.

References

Mikhail Alexandrov, Maxim Kontsevich, Albert Schwarz, and Oleg Zaboronsky. The geometry of the master equation and topological quantum field theory. *Internat. J. Modern Phys. A*, 12(7):1405–1429, 1997.

^[2] Daniel Álvarez. Transitive courant algebroids and double symplectic groupoids. Int. Math. Res. Not. IMRN, 2023.

- [3] Daniel Álvarez. Poisson groupoids and moduli spaces of flat bundles over surfaces. Adv. Math., 440:Paper no. 109523, 2024.
- [4] Camilo Arias Abad and Florian Schätz. The A_{∞} de Rham theorem and integration of representations up to homotopy. Int. Math. Res. Not. IMRN, (16):3790–3855, 2013.
- [5] Camilo Arias Abad and Florian Schätz. Higher holonomies: comparing two constructions. Differential Geom. Appl., 40:14–42, 2015.
- [6] Michael Artin, Alexander Grothendieck, and Jean-Louis Verdier. Théorie des Topos et Cohomologie Etale des Schémas. Séminaire de Géométrie Algébrique du Bois-Marie (SGA 4), volume 269, 270, 305. Lecture Notes in Mathematics. Springer-Verlag, Berlin-New York, 1972-1973.
- [7] Paolo Aschieri, Luigi Cantini, and Branislav Jurčo. Nonabelian bundle gerbes, their differential geometry and gauge theory. Comm. Math. Phys., 254(2):367–400, 2005.
- [8] John C. Baez. Higher Yang-Mills Theory, arXiv:hep-th/0206130, 2002.
- John C. Baez and Alissa S. Crans. Higher-dimensional algebra. VI. Lie 2-algebras. Theory Appl. Categ., 12:492–538, 2004.
- [10] John C. Baez and Aaron D. Lauda. Higher-dimensional algebra. V. 2-groups. Theory Appl. Categ., 12:423–491, 2004.
- [11] John C. Baez and Urs Schreiber. Higher gauge theory. In Categories in algebra, geometry and mathematical physics, volume 431 of Contemp. Math., pages 7–30. Amer. Math. Soc., Providence, RI, 2007.
- [12] Chengming Bai, Yunhe Sheng, and Chenchang Zhu. Lie 2-bialgebras. Comm. Math. Phys., 320(1):149–172, 2013.
- [13] Kai Behrend and Ping Xu. Differentiable stacks and gerbes. J. Symplectic Geom., 9(3):285-341, 2011.
- [14] A. Beilinson and J. Bernstein. A proof of Jantzen conjectures. In I. M. Gelfand Seminar, volume 16, Part 1 of Adv. Soviet Math., pages 1–50. Amer. Math. Soc., Providence, RI, 1993.
- [15] Giuseppe Bonavolontà and Norbert Poncin. On the category of Lie n-algebroids. J. Geom. Phys., 73:70–90, 2013.
- [16] Francesco Bonechi, Nicola Ciccoli, Camille Laurent-Gengoux, and Ping Xu. Shifted Poisson structures on differentiable stacks. Int. Math. Res. Not. IMRN, (9):6627–6704, 2022.
- [17] Olivier Brahic and Cristian Ortiz. Integration of 2-term representations up to homotopy via 2-functors. Trans. Amer. Math. Soc., 372(1):503–543, 2019.
- [18] Lawrence Breen and William Messing. Differential geometry of gerbes. Adv. Math., 198(2):732–846, 2005.
- [19] Ronald Brown and Kirill C. H. Mackenzie. Determination of a double Lie groupoid by its core diagram. J. Pure Appl. Algebra, 80(3):237–272, 1992.
- [20] Andrew J. Bruce. Homological sections of Lie algebroids. Differential Geom. Appl., 79:Paper No. 101826, 21, 2021.
- [21] Andrew J. Bruce, Katarzyna Grabowska, and Janusz Grabowski. Graded bundles in the category of Lie groupoids. SIGMA Symmetry Integrability Geom. Methods Appl., 11:Paper 090, 25, 2015.
- [22] Henrique Bursztyn and Alejandro Cabrera. Multiplicative forms at the infinitesimal level. Math. Ann., 353(3):663-705, 2012.
- [23] Henrique Bursztyn, Alejandro Cabrera, and Matias del Hoyo. Vector bundles over Lie groupoids and algebroids. Adv. Math., 290:163–207, 2016.
- [24] Henrique Bursztyn, Alejandro Cabrera, and Matias del Hoyo. Poisson double structures. J. Geom. Mech., 14(2):151–178, 2022.
- [25] Henrique Bursztyn, Marius Crainic, Alan Weinstein, and Chenchang Zhu. Integration of twisted Dirac brackets. Duke Math. J., 123(3):549–607, 2004.
- [26] Henrique Bursztyn, Miquel Cueca, and Rajan A. Mehta. A geometric characterization of N-manifolds and the Frobenius theorem. *Preprint: arXiv:2402.17746*, pages 1–31, 2024.
- [27] Henrique Bursztyn, Thiago Drummond, and Clarice Netto. Dirac structures and Nijenhuis operators. Math. Z., 302(2):875–915, 2022.
- [28] Henrique Bursztyn, Francesco Noseda, and Chenchang Zhu. Principal actions of stacky Lie groupoids. Int. Math. Res. Not. IMRN, (16):5055–5125, 2020.
- [29] Damien Calaque, Rune Haugseng, and Claudia Scheimbauer. The AKSZ construction in derived algebraic geometry as an extended topological field theory. *Preprint*, arXiv:2108.02473, 2021.
- [30] Alberto S. Cattaneo and Giovanni Felder. A path integral approach to the Kontsevich quantization formula. Comm. Math. Phys., 212(3):591–611, 2000.

- [31] Alberto S. Cattaneo and Giovanni Felder. Poisson sigma models and symplectic groupoids. In *Quantization of singular symplectic quotients*, volume 198 of *Progr. Math.*, pages 61–93. Birkhäuser, Basel, 2001.
- [32] Alberto S. Cattaneo and Florian Schätz. Introduction to supergeometry. Rev. Math. Phys., 23(6):669–690, 2011.
- [33] Alberto S. Cattaneo and Marco Zambon. A supergeometric approach to Poisson reduction. Comm. Math. Phys., 318(3):675–716, 2013.
- [34] Zhuo Chen, Honglei Lang, and Zhangju Liu. The weak Lie 2-algebra of multiplicative forms on a quasi-Poisson groupoid. arXiv:2302.01294, pages 1–34, 2023.
- [35] Zhuo Chen, Mathieu Stiénon, and Ping Xu. Poisson 2-groups. J. Differential Geom., 94(2):209-240, 2013.
- [36] Zhuo Chen, Mathieu Stiénon, and Ping Xu. Weak Lie 2-bialgebras. J. Geom. Phys., 68:59–68, 2013.
- [37] Theodore J. Courant. Dirac manifolds. Trans. Amer. Math. Soc., 319(2):631–661, 1990.
- [38] Marius Crainic. Prequantization and Lie brackets. J. Symplectic Geom., 2(4):579-602, 2004.
- [39] Matias del Hoyo and Cristian Ortiz. Morita equivalences of vector bundles. Int. Math. Res. Not. IMRN, (14):4395–4432, 2020.
- [40] Alfonso Garmendia and Sylvie Paycha. Principal bundle groupoids, their gauge group and their nerve. J. Geom. Phys., 191:Paper No. 104865, 21, 2023.
- [41] Grégory Ginot and Mathieu Stiénon. G-gerbes, principal 2-group bundles and characteristic classes. J. Symplectic Geom., 13(4):1001–1047, 2015.
- [42] Janusz Grabowski. Brackets. Int. J. Geom. Methods Mod. Phys., 10(8):1360001, 45, 2013.
- [43] Janusz Grabowski and PawełUrbański. Lie algebroids and Poisson-Nijenhuis structures. volume 40, pages 195–208. 1997. Quantizations, deformations and coherent states (Białowieża, 1996).
- [44] Alfonso Gracia-Saz and Rajan A. Mehta. VB-groupoids and representation theory of Lie groupoids. J. Symplectic Geom., 15(3):741–783, 2017.
- [45] Markus Hansen and Thomas Strobl. First class constrained systems and twisting of Courant algebroids by a closed 4-form. In *Fundamental interactions*, pages 115–144. World Sci. Publ., Hackensack, NJ, 2010.
- [46] Richard Hepworth. Vector fields and flows on differentiable stacks. Theory Appl. Categ., 22:542–587, 2009.
- [47] Juan Sebastián Herrera-Carmona and Cristian Ortiz. The Chern-Weil Lecomte characteristic map for l_{∞} algebras. *Preprint: arXiv:2306.04465*, pages 1–63, 2023.
- [48] Benjamin Hoffman and Reyer Sjamaar. Stacky Hamiltonian actions and symplectic reduction. Int. Math. Res. Not. IMRN, (20):15209–15300, 2021.
- [49] Johannes Huebschmann. Poisson cohomology and quantization. J. Reine Angew. Math., 408:57–113, 1990.
- [50] David Iglesias-Ponte, Camille Laurent-Gengoux, and Ping Xu. Universal lifting theorem and quasi-Poisson groupoids. J. Eur. Math. Soc. (JEMS), 14(3):681–731, 2012.
- [51] Yvette Kosmann-Schwarzbach. Quasi, twisted, and all that...in Poisson geometry and Lie algebroid theory. In *The breadth of symplectic and Poisson geometry*, volume 232 of *Progr. Math.*, pages 363–389. Birkhäuser Boston, Boston, MA, 2005.
- [52] Bertram Kostant. Quantization and unitary representations. I. Prequantization. In Lectures in Modern Analysis and Applications, III, volume Vol. 170 of Lecture Notes in Math., pages 87–208. Springer, Berlin-New York, 1970.
- [53] Alexei Kotov and Thomas Strobl. Characteristic classes associated to Q-bundles. Int. J. Geom. Methods Mod. Phys., 12(1):1550006, 26, 2015.
- [54] Andreas Kraft and Jonas Schnitzer. An introduction to L_{∞} -algebras and their homotopy theory for the working mathematician. *Rev. Math. Phys.*, 36(1):Paper no. 2330006, 2024.
- [55] Olga Kravchenko. Strongly homotopy Lie bialgebras and Lie quasi-bialgebras. Lett. Math. Phys., 81(1):19–40, 2007.
- [56] Derek Krepski. Groupoid equivariant prequantization. Comm. Math. Phys., 360(1):169–195, 2018.
- [57] Honglei Lang, Yunhe Sheng, and Xiaomeng Xu. Strong homotopy Lie algebras, homotopy Poisson manifolds and Courant algebroids. Lett. Math. Phys., 107(5):861–885, 2017.
- [58] Camille Laurent-Gengoux and Ping Xu. Quantization of pre-quasi-symplectic groupoids and their Hamiltonian spaces. In *The breadth of symplectic and Poisson geometry*, volume 232 of *Progr. Math.*, pages 423–454. Birkhäuser Boston, Boston, MA, 2005.
- [59] David Li-Bland and Eckhard Meinrenken. Dirac Lie groups. Asian J. Math., 18(5):779–815, 2014.
- [60] David S. Li-Bland. LA-Courant algebroids and their applications. ProQuest LLC, Ann Arbor, MI, 2012. Thesis (Ph.D.)–University of Toronto (Canada).

DANIEL ÁLVAREZ AND MIQUEL CUECA

- [61] Zhang-Ju Liu, Alan Weinstein, and Ping Xu. Manin triples for Lie bialgebroids. J. Differential Geom., 45(3):547–574, 1997.
- [62] Jiang-Hua Lu and Alan Weinstein. Groupoïdes symplectiques doubles des groupes de Lie-Poisson. C. R. Acad. Sci. Paris Sér. I Math., 309(18):951–954, 1989.
- [63] Kirill C. H. Mackenzie. Double Lie algebroids and second-order geometry. I. Adv. Math., 94(2):180–239, 1992.
- [64] Kirill C. H. Mackenzie. Double Lie algebroids and the double of a lie bialgebroid. arXiv:math/9808081, pages 1–19, 1998.
- [65] Kirill C. H. Mackenzie. Double Lie algebroids and second-order geometry. II. Adv. Math., 154(1):46–75, 2000.
- [66] Kirill C. H. Mackenzie. General theory of Lie groupoids and Lie algebroids, volume 213 of London Mathematical Society Lecture Note Series. Cambridge University Press, Cambridge, 2005.
- [67] Kirill C. H. Mackenzie and Ping Xu. Classical lifting processes and multiplicative vector fields. Quart. J. Math. Oxford Ser. (2), 49(193):59–85, 1998.
- [68] Ana Carolina Mançur. Drinfeld doubles of Poisson double structures. 2024. PhD-Theis, IMPA.
- [69] João Faria Martins and Roger Picken. Surface holonomy for non-abelian 2-bundles via double groupoids. Adv. Math., 226(4):3309–3366, 2011.
- [70] Rajan A. Mehta. Supergroupoids, double structures, and equivariant cohomology. ProQuest LLC, Ann Arbor, MI, 2006. Thesis (Ph.D.)–University of California, Berkeley.
- [71] Rajan A. Mehta. Q-algebroids and their cohomology. J. Symplectic Geom., 7(3):263–293, 2009.
- [72] Rajan A. Mehta. Q-groupoids and their cohomology. Pacific J. Math., 242(2):311-332, 2009.
- [73] Rajan A. Mehta. On homotopy Poisson actions and reduction of symplectic Q-manifolds. Differential Geom. Appl., 29(3):319–328, 2011.
- [74] Rajan A. Mehta. Representing representations up to homotopy. 2014. Talk slides.
- [75] Rajan A. Mehta and Marco Zambon. L_{∞} -algebra actions. Differential Geom. Appl., 30(6):576–587, 2012.
- [76] Ieke Moerdijk and Janez Mrčun. On integrability of infinitesimal actions. Amer. J. Math., 124(3):567–593, 2002.
- [77] Ieke Moerdijk and Janez Mrčun. Introduction to foliations and Lie groupoids, volume 91 of Cambridge Studies in Advanced Mathematics. Cambridge University Press, Cambridge, 2003.
- [78] Thomas Nikolaus and Konrad Waldorf. Four equivalent versions of nonabelian gerbes. Pacific J. Math., 264(2):355–419, 2013.
- [79] Cristián Ortiz. Multiplicative Dirac structures. Pacific J. Math., 266(2):329–365, 2013.
- [80] Cristian Ortiz and James Waldron. On the Lie 2-algebra of sections of an *LA*-groupoid. J. Geom. Phys., 145:103474, 34, 2019.
- [81] Jonathan P. Pridham. An outline of shifted Poisson structures and deformation quantisation in derived differential geometry. pages 1–43, 2018. arXiv:1804.07622.
- [82] Dorette A. Pronk. Etendues and stacks as bicategories of fractions. Compositio Math., 102(3):243–303, 1996.
- [83] Juan P. Roggiero. Lie Algebroids over Quotient Spaces. 2011. PhD theis.
- [84] Dmitry Roytenberg. Courant algebroids, derived brackets and even symplectic supermanifolds. ProQuest LLC, Ann Arbor, MI, 1999. Thesis (Ph.D.)–University of California, Berkeley.
- [85] Dmitry Roytenberg. On the structure of graded symplectic supermanifolds and Courant algebroids. In Quantization, Poisson brackets and beyond (Manchester, 2001), volume 315 of Contemp. Math., pages 169–185. Amer. Math. Soc., Providence, RI, 2002.
- [86] Albert Schwarz. Geometry of Batalin-Vilkovisky quantization. Comm. Math. Phys., 155(2):249–260, 1993.
- [87] Pavol Ševera. Letters to Alan Weinstein about Courant algebroids. pages 1–29, 2001. arXiv:1707.00265.
- [88] Pavol Ševera. Some title containing the words "homotopy" and "symplectic", e.g. this one. In *Travaux mathématiques. Fasc. XVI*, volume 16 of *Trav. Math.*, pages 121–137. Univ. Luxemb., Luxembourg, 2005.
- [89] Luca Stefanini. On morphic actions and integrability of LA-groupoids. PhD thesis, ETH Zürich, 2008.
- [90] Arkady Yu. Vaintrob. Lie algebroids and homological vector fields. Uspekhi Mat. Nauk, 52(2(314)):161–162, 1997.
- [91] Theodore Voronov. Graded manifolds and Drinfeld doubles for Lie bialgebroids. In Quantization, Poisson brackets and beyond (Manchester, 2001), volume 315 of Contemp. Math., pages 131–168. Amer. Math. Soc., Providence, RI, 2002.
- [92] Jan Vysoký. Graded generalized geometry. J. Geom. Phys., 182:Paper No. 104683, 27, 2022.
- [93] Konrad Waldorf. A global perspective to connections on principal 2-bundles. Forum Math., 30(4):809–843, 2018.
- [94] James Waldron. Lie Algebroids over Differentiable stacks. Ph.D. Thesis, arXiv:1511.07366, pages 1–98, 2015.

- [95] Alan Weinstein and Ping Xu. Extensions of symplectic groupoids and quantization. J. Reine Angew. Math., 417:159–189, 1991.
- [96] Ping Xu. Momentum maps and Morita equivalence. J. Differential Geom., 67(2):289–333, 2004.
- [97] Chenchang Zhu. Lie II theorem for Lie algebroids via stacky Lie groupoids. In *Trends in mathematics*, pages 115–139. Universitätsdrucke Göttingen, Göttingen, 2008.
- [98] Chenchang Zhu. n-groupoids and stacky groupoids. Int. Math. Res. Not. IMRN, (21):4087-4141, 2009.

Departament of Mathematics, University of Toronto, 40 St. George St., Toronto, Ontario, M5S $2\mathrm{E}4$

Email address: dalv@math.toronto.edu

MATHEMATICS INSTITUTE, GEORG-AUGUST-UNIVERSITY OF GÖTTINGEN, BUNSENSTRASSE 3-5, GÖTTINGEN 37073, GERMANY

Email address: miquel.cuecaten@mathematik.uni-goettingen.de